www

How does competition shape managerial decisions? Product market competition and financial statement comparability

Muhammad Ansar Majeed

International Business School, Zhejiang Gongshang University, Hangzhou, China

Chao Yan

School of Accounting, Zhongnan University of Economics and Law, Wuhan, China, and

Muhammad Zubair Tauni

International Business School, Zhejiang Gongshang University, Hangzhou, China

Abstract

Purpose – The purpose of this paper is to explore the effects of competitive pressure on financial statements' comparability (comparability) by analyzing various dimensions of competition.

Design/methodology/approach – The authors study the effect of competition on comparability using the comparability measure of De Franco *et al.* (2011) and various proxies for competition, competition from existing/potential rivals and non-price competition (NPC).

Findings – This study documents that competition is positively associated with comparability, and this effect is more (less) pronounced for industry followers (leaders). The authors also document that competition from existing rivals enhances comparability, but competition from potential entrants does not. Moreover, NPC is also a significant determinant of comparability. Furthermore, the competition from existing/potential rivals plays no significant role in the production of comparable financial statements in state-owned enterprises. The results are robust to alternative measures of comparability and methodological approaches.

Originality/value – This study is the first empirical study that documents a new channel (comparability) through which competition affects financial statements. The findings support the argument that competitive pressure acts as a governance mechanism, disciplines management and increases comparability leading to lower information asymmetry (governance view). However, the findings contest the argument that higher competition motivates managers to withhold information (proprietary cost hypothesis). By examining the effect of state ownership, this study might also help to characterize the effects of changes in corporate objectives on managerial decisions related to financial reporting.

Keywords Competition, Comparability, Managerial opportunism, Proprietary cost hypothesis Paper type Research paper

1. Introduction

Earlier research has recognized the effects of competitive pressure on management decisions, such as cash holding and cash distribution (Alimov, 2014; Byoun and Xu, 2016), investment (Akdoğu and MacKay, 2009, 2012), initial public offerings (Chemmanur and He, 2011; Chemmanur and Yan, 2017) and financial reporting (Li, 2010; Dhaliwal *et al.*, 2014). The study of the effects of product market competition (competition hereafter) on managerial behavior has produced two competing views: competition intensifies agency

JEL Classification — G30, L11, M4

This research was supported by the National Natural Science Foundation of China (Grant No. 71373236). The authors are thankful to Professor Richard A. Hunt (associate editor) and anonymous reviewers for their guidance and constructive suggestions. The authors appreciate the workshop participants at the School of Accounting and seminar participants at the International Business School, Zhejiang Gongshang University for their comments, and also thank Dr Najid Ahmed for his suggestions.

C

Management Decision Vol. 56 No. 11, 2018 pp. 2437-2471 © Emerald Publishing Limited 0025-1747 DOI 10.1108/MD-04-2017-0319



2437

Received 9 April 2017 Revised 4 January 2018 15 May 2018 Accepted 28 May 2018

Managerial decisions

problems and managerial opportunism (governance view), according to one school of thought. Earlier research has provided theoretical grounds for competition being a strong force that can act as a governance mechanism and curtail managerial opportunism, thereby reducing agency conflicts (Alchian, 1950; Stigler, 1958; Hart, 1983; Scharfstein, 1988). These studies suggest that competition acts as a strong force, which aligns the interests of management with shareholders. The preceding literature further suggests that competition increases the information available to principles, enabling them to effectively monitor
managerial activities (Holmström, 1982; Hart, 1983; Nalebuff and Stiglitzs, 1983; Schmidt, 1997; Baggs and Bettignies, 2007). However, another school of thought argues that competition intensifies agency problems and managerial opportunism (Datta *et al.*, 2013; Horn *et al.*, 1994; Scharfstein, 1988). This line of study subscribes to the "dark side of competition" view, i.e. competition promotes unethical behaviors, such as child labor, excessive executive compensation, earnings manipulation, etc. (Shleifer, 2004).

Similarly, the preceding studies have also provided competing views regarding the impact of competition on financial reporting. One stream of studies suggests that competition acts as a governance mechanism, disciplining management and ultimately enhancing financial reporting transparency (i.e. lower information asymmetry), thereby improving the information environment (Li, 2010; He, 2012; Laksmana and Yang, 2014; Majeed and Zhang, 2016), while the other stream of research argues that firms, when competition is higher, prefer an opaque information environment and disclose less information because managers attempt to shield proprietary information (Verrecchia, 1983; Stivers, 2004; Verrecchia and Weber, 2006; Ali *et al.*, 2014). Moreover, a positive association between information asymmetry and competition might also arise because of higher managerial opportunism, as competitive pressure reduces the return on managerial efforts, and management is forced to misstate the true economic performance (Karuna, 2007). In light of these arguments, competing views have arisen out of the nexus of competition and financial reporting. These competing arguments motivated our study of the influence of competition on comparability.

Financial statement comparability (comparability hereafter) is an important and distinctive feature of financial reporting. Comparability improves the utility of accounting information. It is a qualitative aspect of financial reporting that improves information quality and reduces information uncertainty and information asymmetry. Comparability helps in making "like things looking alike and different things looking different" (Barth, 2013). In economic decision making, the financial performance of firms cannot be viewed in isolation because all of the alternatives should be compared for optimal decisions. Comparability enables the users of financial statements to identify the similarities and differences in economic performance. Comparability is a key and valuable construct for investors because all of the decisions related to investments and lending are made after investors and lenders compare alternatives. Such a comparison is unlikely if comparable information is not available (Financial Accounting Standards Board (FASB) 2010). FASB (2010) stated, "comparability is the quality of information that enables users to identify similarities and differences between two sets of economic phenomena."

Prior research has exhibited numerous benefits of accounting comparability. According to De Franco *et al.* (2011), higher comparability also increases the availability of higher quality information. They also suggested that comparable information reduces information acquisition and processing costs and improves the quality and quantity of information available regarding firms. Kim *et al.* (2013) proposed two potential benefits of comparability. They argued that comparability reduces information asymmetry and information uncertainty, thereby improving the information processing ability of less informed investors and consequently reducing credit risk. Habib *et al.* (2017) suggested that comparability reducing of managerial activities easier and



2438

MD

facilitates the efficient allocation of capital resources. Other benefits include lower expected stock price crash risk (Kim *et al.*, 2016), improved firm value, higher liquidity, greater forecast accuracy and forecast agreements (Neel, 2017), higher informativeness of stock price (Choi *et al.*, 2017) and lower underpricing around seasoned equity offerings (Shane *et al.*, 2014). All of these benefits of accounting comparability arise from lower information asymmetry and information uncertainty.

The immense significance of comparability as a qualitative feature of accounting makes it crucial for comprehending the factors that determine this attribute. Despite the significance of comparability highlighted by policy makers and the preceding research, the literature on the determinants of comparability is fairly scarce. The majority of prior research into the determinants of comparability has fixated on the role played by accounting standards (e.g. IFRS) and has ignored other business and institutional factors that shape financial reporting (Barth et al., 2012; Lang et al., 2010). It is important to note that financial reporting outcomes are not shaped by accounting standards only because of the vital roles played by institutional and economic factors (Ball et al., 2003; Leuz et al., 2003). Barth (2013) suggested that the adoption of "global accounting standards is necessary but not sufficient step to achieving comparability." These premises inspired us to study of the role that competition plays in the production of comparability. The contradictory views regarding the impact of competition on financial reporting also motivated this study. Moreover, comparability is associated with information asymmetry, and prior studies (e.g. Leuz, 2003) employed information asymmetry as an indicator of accounting quality. We argue that competition potentially influences comparability. This association is expected because competition results in higher (lower) information asymmetry because of higher (lower) agency costs or higher proprietary costs, and comparability reduces information asymmetry and information uncertainty. Specifically, this study addresses the following questions: How does competition influence comparability? What is the effect of competition from existing rivals (potential entrants) on comparability? What is the impact of government ownership on the relationship between competition (from existing and potential rivals) and comparability? This study sets the two following objectives. First, we study the role of competitive pressure in affecting managerial decisions by analyzing the impact of competition on qualitative aspects of financial statements, i.e. comparability. Second, we examine the role played by government ownership on the relationship between product market competition and comparability.

Our study contributes to the literature by documenting the impact of competition on comparability. We document that competition is an important determinant of comparability. The results suggest that competition acts as a disciplinary mechanism, reducing managerial opportunism and agency conflicts and consequently improving accounting quality by increasing comparability. Although competition increases comparability, we do not find such an association for industry leaders, consistent with the view that industry leaders face less competitive pressure than industry followers (Dhaliwal *et al.*, 2014; Majeed *et al.*, 2017). Our results further suggest that competition from existing rivals increases comparability, but competition from potential entrants has no significant relationship with comparability. Moreover, we also suggest that non-price competition (NPC) positively influences comparability. Our study further explores the effect of state ownership on the relationship between competition and comparability. We document that competition from existing, as well as potential, rivals plays no statistically significant role in increasing (decreasing) comparability for state-owned enterprises (SOEs).

Our study contributes to the literature on comparability and competition in various ways. First, to the best of our knowledge, this study is the first hypothesizing and documenting the effect of competition on the production of comparability within a country. Prior studies have documented the effects of comparability, but literature on the



Managerial decisions

MDdeterminants of comparability is lacking. The preceding research has mainly focused on the
effects of accounting standards (i.e. IFRS adoption) on the production of comparability.
However, accounting standards are not the only factor that shapes comparable financial
statement decisions (Leuz, 2003; Barth, 2013). Our study extends this argument and offers
evidence that competition plays a significant role in the production of comparable financial
statements. This study, unlike previous studies of the nexus of competition and financial
reporting, examines the effect of competition on qualitative aspects of financial reporting.
In this way, our study proposes a new channel through which competition affects financial
reporting, i.e. comparability.

Second, our findings provide empirical evidence for the argument that competition disciplines management, decreases agency conflicts (Baggs and Bettignies, 2007) and improves the financial reporting environment. However, this study contests the proprietary cost hypothesis (Verrecchia, 1983, 1990; Verrecchia and Weber, 2006; Ali *et al.*, 2014) and the "dark side of competition" view (Shleifer, 2004; Lin *et al.*, 2015). Third, we exploit the unique Chinese institutional environment, which is characterized by the presence of a large number of SOEs. These SOEs have different objectives and different agency conflicts (Faccio, 2006; Ali *et al.*, 2007; Majeed *et al.*, 2018). Therefore, we examine the effects of two dimensions of competition, i.e. competition from existing rivals and competition from potential entrants, on the financial statement comparability of SOEs. Our study could increase understanding of the effects of changes on corporate objectives and agency conflicts regarding managerial behavior. Fourth, since competition reduces agency slack and disciplines management, our study might be useful for other stakeholders, particularly investors, in understanding the link between competition and comparability through information asymmetry and information uncertainty.

The remainder of the paper is organized as follows. Section 2 provides institutional background on China, while Section 3 provides the hypothesis development. Section 4 explains the variable measurements and research design. Section 5 provides the results. Section 6 includes additional tests and robustness checks, while Section 7 presents the paper's conclusions.

2. Institutional background

China is quite different from other developed and developing economies because its institutional and legal environment is quite different. Therefore, to contextualize the association between competition and comparability, an overview of the Chinese corporate setting is crucial.

2.1 Product market competition in China

After the establishment of People's Republic of China in 1949, China became a centrally planned economy. The government essentially made all decisions regarding output (production) targets, prices, and resource allocation. Foreign investment and private business were virtually non-existent, and the market was captured by SOEs (Morrison, 2015). Actually, there was no competition among firms until 1978. However, after 1978, a series of economic reforms resulted in lowering of entry barriers for private firms, elimination of price controls and freeing of internal trade. Moreover, China adopted a policy promoting foreign trade and investment (Holz, 2008). An increasing number of private businesses were started, and the sectors of the economy once monopolized by state-owned companies now have numerous private firms as well. The result of increased private sector involvement was that almost 90 percent of retail prices were entirely determined by the market until 2008 (Conway *et al.*, 2010). The reforms of 1980s and 1990s also increased foreign trade and investment (Cai and Liu, 2009).



When China joined the World Trade Organization (WTO) in 2001, the average import tariffs decreased, generating import shock and thereby increasing competition in domestic markets (Tybout, 2004). A study by Koopman et al. (2011) examined the share of Chinese exports produced in China actually. They reported that, before joining the WTO, almost 50 percent of Chinese exports were actually imported and were then exported after value addition. Until 2007, the foreign share was almost 60 percent of Chinese exports, suggesting that imports increased more than the exports. Furthermore, China introduced a new antimonopoly law, which also helped to increase competition (Owen et al., 2008). Over the last three decades, China has moved from a centrally planned economy to a market-based economy. Although market forces play an increasingly important role in determining economic behavior, the transition is still incomplete (Chen et al., 2014). Today, private companies, foreign businesses and SOEs compete with each other, but still there is uneven competition across industrial sectors. Although certain industries face stiff competition, along with significant barriers to entry for new entrants, some other industries, such as telecommunications and petroleum, are still monopolistically under central or local government control. According to Jiang et al. (2015), there is a "high degree of variation in product market competition within industries across time and across industries within each time period" in China. Such an environment provides us with an interesting institutional background to study the impact of competitive pressure on comparability.

2.2 Corporate information environment in China

Higher quality public disclosure decreases information asymmetry and results in higher valuation, lower cost of equity and well-functioning capital markets. However, emerging economies, because of weak institutional environments and investor protection, have opaque information environments. China is unique in the sense that, despite being the second-largest economy, the Chinese corporate information environment is among the lowest ranked in the world. Piotroski and Wong (2012) suggested that incentives induced by the legal, political and cultural background are to be blamed for the opaque corporate information environment in China. For example, the listing regulations provide incentives to distort the true economic performance of firms. Per listing regulations, a firm is delisted if it reports loss for three consecutive years. Such rules and regulations provide motivations to inflate earnings and resultantly increase information asymmetry. The prior literature has provided empirical support for the impact of listing regulations on the distortion of the information environment. Peng et al. (2011) reported that the controlling shareholders in financially healthy (distressed) firms are more likely to conduct related party transactions for the tunneling (propping up) of their listed firms. The listed firms in China face two types of risk, i.e. they can be delisted or lose the right to issue new shares. Because listing is a lengthy and difficult process in China, there are strong incentives for the controlling shareholders to prop up listed firms in cases of distress so that they can continue to enjoy controlling the privileges of listed firms or access the financial markets in the long run (Peng et al., 2011).

The International Financial Reporting Standards (IFRS) were adopted worldwide to improve the corporate information environment. China also converged its domestic accounting standards with the IFRS for the same purpose; however, this convergence has not produced the desired results (He *et al.*, 2012). In addition, capital market pressures to reduce the information asymmetry are limited because a major source of financing is still the banking sector (Allen *et al.*, 2012). Furthermore, the short investment horizon of investors is also a reason for limited capital market pressures. The Chinese equity market is known for exceptionally high trading turnover, indicating that investors have a short-term investment horizon. Not only individual investors but also institutional investors, contrary to the trend, also have short-run investment horizons (Jiang and Kim, 2015).



Managerial decisions

MD 56,11 Investor sophistication also differentiates the Chinese equity market from other markets in developed countries, where individual investors hold a large number of tradable shares. Usually, the big auditors are associated with improved quality of accounting information because they play a monitoring role. However, the big 4 auditors provide lower quality audits for the firms listed in China, and the weak institutional environment of China is the main culprit for this lower audit quality (Ke *et al.*, 2015).

Other important characteristics of the Chinese environment include government interference in the corporate sector and the importance of government connections for firms. Chinese government intervention can occur through two channels. One channel is by direct ownership of corporate firms, while the other channel is the strategic selection of special economic development areas (Shao *et al.*, 2015). Moreover, the majority of the listed firms in China are directly or indirectly controlled by the state, as two thirds of listed companies were state-owned at the end of 2013 (Wu *et al.*, 2016). SOEs enjoy a lower cost of debt (Shailer and Wang, 2015) and have higher principle–principle agency conflicts, i.e. conflict of interests between controlling and non-controlling shareholders (Ali *et al.*, 2007). These SOEs are regarded as less risky by lenders (Chen *et al.*, 2010) because they can be bailed out by the government in cases of financial distress (Faccio, 2006). The SOEs are also given preferential treatment in many ways. For example, NSOEs, compared to SOEs, face strict institutional regulations (e.g. loan restrictions or industrial barriers). Therefore, preferential treatment, better access to finance, government support in cases of distress and different objectives from NSOEs lead to different disclosure choices in SOEs.

3. Literature review and hypothesis development

3.1 Product market competition and comparability

Financial statements are prepared with the aim of increasing the availability of financial information for stakeholders. In particular, the separation of ownership and control creates agency conflicts resulting from information asymmetry between management and owners (Jensen and Meckling, 1976). Healy and Palepu (2001) suggested that agency problems and information asymmetry create demand for financial disclosures. However, corporate governance facilitates the alignment of managerial interests with the shareholders, thereby resulting in the production of more reliable financial reports, which, in turn, improve the information environment (Watts and Zimmerman, 1986). Some recent studies (Kanagaretnam et al., 2007; Holm and Schøler, 2010; Han et al., 2014) have also noted that effective governance mechanisms largely improve the information environment and hence reduce information asymmetry. Moreover, the reliability of the financial information decreases in the presence of managerial opportunism, which can be controlled through some governance mechanisms (Dechow et al., 1996; Wild, 1996). Several incidents in the past, such as the bankruptcies of Enron and Worldcom, have demonstrated the importance of the financial reporting environment. Prior studies have also suggested that corporate governance is more important when likely agency costs are higher (Chi and Lee, 2010) (such as Chinese institutional settings).

Comparability is a qualitative aspect of financial reporting with immense importance, as it increases the usefulness of accounting information. The immense importance of comparability has also been emphasized in accounting textbooks (e.g. Phillips *et al.*, 2013). It enables the users of financial reports to compare similarities and differences and to improve decision making by stakeholders (e.g. lenders, investors, creditors and even regulators). The preceding studies have documented various benefits of comparability. For example, a higher level of comparability improves analyst forecast accuracy and reduces information asymmetry (De Franco *et al.*, 2011), and it lowers credit risk by decreasing information asymmetry and information uncertainty, thereby improving information processing capabilities (Kim *et al.*, 2013). The lower information asymmetry resulting from higher comparability produces



numerous other benefits as well, such as lower cost of capital (Shane *et al.*, 2014), efficient capital allocation (Barth, 2013; Chen *et al.*, 2015), availability of more firm-specific information (Choi *et al.*, 2017) and higher firm value (Neel, 2017). Kim *et al.* (2016) argued that higher comparability makes information acquisition easier, and the cost to withhold bad news dwarfs the benefits, leading to a lower expected stock price crash risk.

Earlier studies have shown two channels through which competition can affect managerial decisions, such as financial reporting. One channel suggests that competitive pressure acts as a governance mechanism that disciplines the management and reduces the agency cost (governance view). However, another view suggests that competition increases proprietary costs and/or exacerbates agency conflicts (proprietary cost hypothesis). The "governance view" suggests that competition reduces agency problems and aligns the interests of management and shareholders. Earlier research has provided theoretical grounds that competition is a strong force that can act as a governance mechanism and curtail managerial opportunism, thereby reducing agency conflicts (Alchian, 1950; Stigler, 1958; Hart, 1983; Scharfstein, 1988). These models predict that competition is a strong force that aligns the managerial interest with shareholders. The prior literature has suggested that competition increases the information available to principles, enabling them to effectively monitor managerial activities (i.e. agents) (Holmström, 1982; Hart, 1983; Nalebuff and Stiglitzs, 1983). Competition enhances the managerial incentives to select an "optimal governance mechanism" (Demsetz and Lehn, 1985) to evade the liquidation risk (Schmidt, 1997). Nickel (1996) suggested that competition is the "driving force" behind numerous policy changes worldwide.

Some other recent studies have also subscribed to the governance view. These studies suggested that competition increases the likelihood of cash distribution to shareholders (Grullon and Michaely, 2007) and aligns not only the interests of managers and shareholders but also the interests of majority and minority shareholders (He, 2012). Guadalupe and Pérez-González (2010) proposed that the disciplinary role of competition is more profound in weak legal settings (such as China). Some of the studies have even argued that competition substitutes for other disciplinary mechanisms (Chou *et al.*, 2011). Mnasri and Ellouze (2015) also suggested that competition improves governance, reduces agency problems and enhances productivity. Chen *et al.* (2014) also subscribed to the governance view, as they reported a negative relationship between competition and the cost of equity in China.

Prior studies of the nexus of competition corporate information environment have concluded that competition is key determinant of financial reporting (Li, 2010; Datta *et al.*, 2013; Dhaliwal *et al.*, 2014). These studies have suggested that competition limits misleading earnings management (Marciukaityte and Park, 2009), curtails real earnings management (Laksmana and Yang, 2014) and reduces the frequency of accounting restatements (Balakrishnan and Cohen, 2014) by disciplining management. Similarly, recent studies in Chinese institutional settings have also reported a positive relationship between competition and financial reporting quality (Majeed and Zhang, 2016; Majeed *et al.*, 2017).

These studies suggested that increased competition acts as a governance mechanism, aligns the interests of management with the shareholders (i.e. disciplines the management) and consequently improves the corporate information environment. We present the following arguments in favor of our expectations. First, if competition improves the corporate information environment, then comparability, being a qualitative aspect that improves the financial reporting environment (by reducing information asymmetry and information uncertainty), should also increase. Another argument for the association between competition and comparability comes from the fact that, if competition disciplines management and improves information environment, to reap capital market benefits (e.g. lower cost of capital and lower frequency of frequent crashes), it should increase comparability because comparability reduces information acquisition and processing costs



Managerial decisions

- and hence increases capital market benefits. This argument is in line with Shane *et al.* (2014), who suggested that comparability also reduces the cost of capital, indicating that competition might improve comparability to reduce the cost of capital. Based on this discussion, we propose our first hypothesis:
 - *H1a.* There is a positive relationship between product market competition and financial statement comparability.

Another viewpoint (proprietary cost hypothesis) suggests that competition decreases the quality of financial reporting and increases information asymmetry and uncertainty regarding firm performance. Verrecchia (1983) argued that managers in highly competitive industries prefer less informative disclosures to conceal proprietary information, leading to an opaque information environment to reduce predatory threats from rivals. Harris (1998) reported a lower likelihood of separate segment disclosures when competition is high. Competitive pressure also compels managers to withhold proprietary information because disclosures can result in loss of market share (Verrecchia and Weber, 2006). Guo et al. (2015) argued that the benefits and costs of higher quality disclosure differ with competition, and the relationship between competition and earnings quality (EQ) is mainly driven by the variation in benefits and costs. Ali et al. (2014) advanced this line of inquiry and reported that competition is associated with management forecasts, which are not only less frequent but also of shorter time horizons. They also documented that when such firms disclose, they are less prompt in disclosure, indicating that they attempt to delay the information as much as possible to render it less useful. They further argued that firms in concentrated industries choose private placement instead of seasoned equity offerings because private placement does not require comprehensive public disclosures, information is disclosed to only a limited number of investors, reducing the magnitude of leakage of proprietary information, and such private placement deals are publicized after the deal. Ali et al. (2014) also posited that firms in concentrated industries have lower disclosure ratings by analysts, higher analyst forecast dispersion, greater forecast errors and more volatile revisions. They argued that inferior-quality disclosure by firms in concentrated industries can be attributed to the proprietary costs associated with higher quality disclosures in industries facing stiff competition.

The preceding research has also suggested that stiff competition increases managerial opportunism. Similarly, Rotemberg and Scharfstein (1990) demonstrated that competition escalates managers' tendency to manipulate financial information to signal better future performance. Competition reduces managerial efforts to improve efficiency because of the lower benefits of marginal increases in firm efficiency (Martin, 1993; Horn et al., 1994). Shleifer (2004) proposed the "dark side of competition view," i.e. competitive pressure promotes unethical behavior among managers, increases managerial opportunism, reduces the quality of financial reports and encourages an opaque information environment, leading to higher information asymmetry and greater information uncertainty. Karuna (2007) also suggested that competition results in close monitoring of CEO performance, which can lead to higher agency conflict, thereby compromising disclosure quality. Lee and Liu (2016) suggested that an exogenous increase in competition increases the prospects of earnings management and accounting frauds. Lin et al. (2015) also concurred with the "dark side" view of competition. They further argued that predation risk (i.e. the risk of losing market share and profitability) was the key driver of this behavior. Kim and Kim (2017) also argued that competition destroys the information environment by impeding the effectiveness of internal control.

We argue that if managers conceal their proprietary information, they render financial statements less comparable to increased information asymmetry (uncertainty) and information acquisition costs for the following reasons. First, higher information asymmetry



MD

56.11

(uncertainty) and information acquisition costs would not only make it difficult for existing rivals to evaluate the real financial position, but they would also render entry decisions difficult for potential entrants. Therefore, competition can reduce comparability. Although lower information asymmetry (uncertainty), arising from higher comparability, can result in a lower cost of capital (Cuadrado-Ballesteros *et al.*, 2016), prior studies (e.g. Ali *et al.*, 2014) have argued that firms do not choose these financing options, which require comprehensive disclosure. Thus, the managers might not be interested in comparability for their capital market benefits when competition is higher. Second, if the competition induces managerial opportunism (dark side of competition), then higher opportunism can result in lower comparability because managers would attempt to distort the true economic performance of firms by producing less comparable financial statements. All of this discussion leads to our second alternative hypothesis (Figure 1):

H1b. There is a negative relationship between product market competition and financial statement comparability.

3.2 Industry leaders (followers) and comparability

The preceding studies have also examined industry leaders and followers facing competitive pressure, but the intensity of competitive pressure varies for leaders and followers. These studies suggested that the intensity of competition is perceived differently by industry leaders and followers. Therefore, the effects of competition on managerial behavior are also different for industry leaders and followers. Frésard and Valta (2011) and Jiang *et al.* (2015) noted that competitive pressure influences corporate policies (such as capital investment, R&D investment, financing and cash holdings), but the effect is stronger for industry followers. Nickell *et al.* (1992) and Nickel (1996) documented that industry followers face greater competitive pressure because of predation risk. Therefore, industry leaders, compared to industry followers, are less vulnerable to the threats posed by competitive pressure.

A similar argument was also shared by some recent studies examining the effect of competitive pressure on financial reporting (Li, 2010; Dhaliwal *et al.*, 2014). Li (2010) suggested that the effect of competitive pressure on disclosure quality is less pronounced for industry leaders. Li documented that the effect of competitive pressure in reducing profit forecast optimism and investment forecast pessimism is stronger for industry followers, leading to more accurate disclosure. Dhaliwal *et al.* (2014) also suggested that industry followers recognize bad news earlier than good news (a strategic consideration), as these industry followers face greater competitive pressure.

Another argument is that industry leaders, being more visible, can attract "more investor attention" more easily. However, followers must attract investors by other means, such as



Managerial decisions

good corporate governance, lower information asymmetry, etc. Therefore, the incentives of higher quality financial reporting are much greater for industry followers than for industry leaders. The greater capital market incentives of higher quality financial reporting can compel industry followers to improve financial reporting quality by reducing information asymmetry through higher comparability.

The greater competitive pressure faced by industry followers acts as a disciplinary mechanism that reduces agency conflicts and information asymmetry as well (Majeed *et al.*, 2017). Furthermore, lower visibility of industry followers also compels them to decrease information asymmetry to attract investors. Therefore, we expect a positive relationship between industry followers and comparability. Moreover, the industry leaders in most industries (in Chinese institutional settings) are SOEs (Chen *et al.*, 2014) with easy access to financing, enjoying more favorable policies from the government and facing fewer financial constraints. Therefore, these industry leaders have fewer incentives to reduce information asymmetry, i.e. to increase comparability. Hence, a less pronounced relationship is expected between industry leaders and comparability. This discussion leads to the following hypothesis:

H2. The relationship between competition and comparability is stronger (weaker) for industry followers (leaders).

3.3 Competition from existing/potential rivals and comparability

According to Li (2010), firms face two unique dimensions of competition, i.e. competition from potential entrants (entry of new firms to the market affects the incumbents' profits adversely) and competition from existing rivals (firms in the same industry producing the same types of goods threaten the market positions of incumbent firms). These two dimensions of competition influence financial reporting decisions (Li, 2010; Dhaliwal et al., 2014; Majeed and Zhang, 2016). Clinch and Verrecchia (1997) modeled a "post-entry duopoly game" and suggested that more disclosure only limits the benefits of competitive advantages, as firms' attempt to exploit incorrect production decisions. Hence, competition from existing rivals lowers financial reporting benefits, while Darrough and Stoughton (1990) modeled a "two player entry game" and proposed that disclosing valuable information increases capital market benefits but also increases the threat of new entrants when entry costs are lower. Thus, incumbents not only disclose good news (to reap capital market benefits) but also disclose bad news (to deter the threat of new entrants). Therefore, more information disclosure is associated with competition from potential entrants. These arguments suggest that both dimensions of competition affect financial reporting.

Gigler (1994) and Evans and Sridhar (2002) suggested that truthful financial reporting is lacking when product and capital markets are isolated. However, both markets use the same set of financial reports. Therefore, the probability of higher quality disclosure increases in the wake of off-setting demands. The likelihood of overly optimistic disclosures (to reap capital market benefits) decreases when there is a threat of overproduction by competitors for entry games and post-entry games (Gigler, 1994; Evans and Sridhar, 2002), leading to improved financial reporting. The models of Gigler (1994) and Evans and Sridhar (2002) predict that competition from existing, as well as potential, rivals improve the quality of financial disclosures. Similarly, Li (2010) argued that the quality of disclosure might increase in both cases (i.e. competition from existing and potential rivals). Dhaliwal *et al.* (2014) also noted the positive effects of both forms of competition on financial reporting conservatism. The underlying conclusion from these arguments is that the both dimensions, i.e. competition from existing and potential rivals, increase the quality of disclosure. As discussed previously, comparability is a qualitative aspect of financial reporting;



MD

56.11

therefore, the effects of these dimensions of competition on comparability are expected to be positive. Based on these premises, we propose the following hypotheses:

H3a. The competition from existing rivals is positively associated with comparability.

H3b. The competition from potential rivals is positively associated with comparability.

4. Sample selection and research design

4.1 Sample description

Our sample comprises all A-listed non-financial firms for the period spanning from 2005-2014. A-shares are those traded on the Shanghai and Shenzhen stock exchanges in local currency, i.e. Chinese yuan. The data contain 1,810 unique firms. We also omit firms designated as ST and PT. The stocks of ST and PT firms are restricted because of severe financial issues faced by these firms. The data set contains variables form 2005 because we require quarterly data for the measurement of comparability. The data are acquired from the China Stock Market and Accounting Research (CSMAR) database. We obtain all of the data for the measurement of dependent, independent and control variables from the CSMAR database. We omit industries with fewer than 15 firm-year observations for the calculation of comparability because we require top 4/top 10 ranks for our comparability measurement. We define the firms' industries based on the second level classifications of China's security regulatory commission (CSRP). All of the continuous variables are winsorized at 1 and 99 percent of their empirical distributions.

4.2 Model

We employ the following model to examine the relationship between product market competition and financial statement comparability. To test numerous propositions, various structures of this model are used:

 $Comp_{it} = \beta_0 + \beta_1 PMC_{it} + \beta_2 SOE_{it} + \beta_3 EQ_{it} + \beta_4 Volatility_{it} + \beta_5 Size_{it} + \beta_6 MB_{it}$

$$+\beta_7 \text{Lev}_{it} + \beta_8 \text{ROA}_{it} + \text{Firm} + \text{Year} + \varepsilon_{it},$$

where "Comp" is a financial statement comparability measure defined by De Franco *et al.* (2011). PMC represents the intensity of product market competition. A positive coefficient of PMC indicates that higher competition would lead to higher comparability and vice versa. Our control variables include state ownership (SOE) and EQ based on discretionary accruals management using a modified Jones model as described by Kothari *et al.* (2005). Size is the natural log of total assets; MB is the ratio of market to book value of equity; Volatility is the standard deviation of sales; Lev represents the leverage ratio; ROA represents return on assets; and Ind_Dir exhibits the board independence, i.e. the ratio of independent directors to the total number of directors on board. The terms Year and Firm represent year- and firm-fixed effects, respectively.

4.3 Measurement of product market competition

We use Herfindahl–Hirschman index (HHI) to measure competition. The HHI has been widely used as proxy of competition in abundant industrial organization and accounting research (Valta, 2012; Zou *et al.*, 2015). We calculate the HHI using the following formula:

$$H_j = \sum_{i=1}^I X^2_{ij},$$

where X is the market share of firm "i" in industry "j," and market share is calculated

2447

Managerial

decisions

as the net sales of the firm divided by the net total sales of the industry to which the firm belongs: 56.11

X = Market Share of Firm = $\frac{\text{Total Sales of Firm}}{\text{Total Sales of Industry}}$.

For ease of explanation, we multiply " H_i " by -1 to create our proxy for product 2448market competition, i.e. PMC, so that higher values of PMC denote higher competition and vice versa.

4.4 Measurement of competition from existing/potential rivals

Following previous studies (Li, 2010; Dhaliwal et al., 2014; Majeed et al., 2017), we calculate competition from existing and potential rivals. We use principle component analysis and calculate measures of competition from existing and potential rivals. The measure of competition from existing rivals (PMC EXST) is the negative of the first component generated from four variables, i.e. the HHI which is sum of the square of market share of all firms in an industry, the four firm concentration ratio (C4) which is defined as sum of the market shares of the four largest firms in an industry, the number of firms in an industry (NF) and the product market size (MKTS), measured as the natural log of total industry sales. We measure our proxy for competition from potential entrants (PMC POTN) as the negative of the first component generated from four variables, i.e. weighted average PPE (plant, property and equipment), weighted average capital expenditures (CAPX), product market size (MKTS) and industry ROA (ROA), following Dhaliwal et al. (2014) and Li (2010). Although studies have used research and development expenditures (R&D), we do not use it for two reasons. First, our sample starts in 2005, but the disclosure of R&D in China started in 2007. Moreover, the disclosure of R&D is incomplete and mostly missing for a large number of firms during the sample period. Thus, because of missing and incomplete disclosure of R&D expenditure, we do not use these data in our measurement of competition from potential entrants. We use industry ROA because the profitability of an industry is also a vital factor considered by potential entrants, as the profitability exhibits the apparent benefits of entering the market (Darrough and Stoughton, 1990; Newman and Sansing, 1993). Principal components are rotated using the orthogonal rotation technique. Two components are retained after orthogonal rotation. Following Li (2010), we retain components with eigenvalues greater than 1. Table AI provides the details of the measurement of competition from existing/potential rivals.

4.5 Measurement of financial statement comparability

FASB describes comparability as an important construct that enables investors to make informed decisions. Preceding studies have argued that there are two ways to measure comparability. The first method used to measure comparability is the use of (common) similar accounting methods by all firms. This method is called an input-based comparability measure because it requires data collection regarding individual firms' accounting choices and then comparing them with the accounting choices of other firms.

However, previous studies have discussed in detail comparability measures, which, compared with "accounting choices of the firms" (input-based comparability measures), face numerous challenges (De Franco et al., 2011; Barth, 2013; Fang et al., 2017). The first of these challenges is the decision of which accounting choice to use. The second challenge is how to weight such accounting choices. Third, for a large sample size, it is quite difficult (or costly) to collect data for a wide-ranging set of accounting choices. Furthermore, quantifying the degree of implementation of such choices is a quite burdensome task. Not only are the measurement and weighing of such accounting choices difficult, but quantifying the degree



MD

of implementation of such choices is also quite a burdensome task. The implantation problem is particularly important in our (Chinese) setting because of weak enforcement of rules and regulations in China.

Furthermore, using the same set of rules or methods (also called the common method) might be less informative about individual firms' fundamentals, compared to firm-specific methods (Fang *et al.*, 2017). Barth (2013) argued that using same methods (rules) does not constitute comparability. She suggested that using the same set of rules, methods, content and structure by a firm over the period of time is called consistency. Similarly, using similar rules, methods, and contents of financial statements by various firms is called uniformity. Barth (2013) further argued, "uniformity can make unlike things look alike, which impairs, not enhances, comparability." Input-based comparability measures, due to the aforementioned issues/criticism, are a sub-optimal optimal choice.

However, more recent studies have focused on output-based comparability measures. The measure is an output-based comparability measure because it does not rely on firms' actual choices of accounting measurements. The output-based measures label the firms as having comparable financial statements if they produce similar financial statements in response to given economic events. De Franco *et al.*'s (2011) (output-based) comparability measure is very popular and widely used. We use the output-based comparability measure prescribed by De Franco *et al.* (2011) following the preceding studies. De Franco *et al.* (2011) suggested that their measure of comparability is "firm specific, dynamic and capturing similarities over time." We employ an output-based comparability measure because of difficulties and criticism associated with input-based comparability measures. Following De Franco *et al.* (2011), we construct a firm-specific, output-based measure of comparability. De Franco *et al.* (2011) defined their accounting system as "mapping from economic events into financial statements":

Financial Statement_i =
$$f_i$$
(Economic Event_i), (1)

where " f_i " represents the accounting system of firm "i," and the economic event is represented by stock returns. For two firms, to have comparable accounting systems, their mapping of economic events should be similar. Following De Franco *et al.* (2011), we assume " f_i " to be a linear function, as presented in the following equation. To measure the accounting function of a specific firm "i" in each year, we estimate the following time-series regression using 16 previous quarters of data:

$$Earnings_{it} = \alpha_i + \beta_i Return_{it} + \varepsilon_{it}.$$
 (2)

Here, the stock returns are used as a proxy for economic events and an earnings proxy for financial statement outcomes. Earnings represent income before extraordinary items divided by market value of equity at the beginning of the period, while Return is the stock price return during the quarter. We estimate the accounting function for firm "i" and accounting function for firm "j" for a given fiscal year. Comparability between two firms increases with the closeness of the functions between the two firms. When two firms experience the same economic event, there is more comparable accounting functions of firm "i" and the firms produce similar financial statements. The estimated accounting functions of firm "i" are used to predict their respective earnings, assuming that they have experienced the same economic event (i.e. return of firm "i"):

$$E(\text{Earnings})_{iit} = \hat{\alpha}_i + \hat{\beta}_i \text{Returns}_{it}, \qquad (3)$$

$$\mathcal{E}(\text{Earnings})_{ijt} = \widehat{\alpha}_j + \widehat{\beta}_j \text{Returns}_{it}, \tag{4}$$

Managerial decisions

MD 56.11

2450

where $E(\text{Earnings})_{iit}$ denotes firm *i*'s predicted earnings given the accounting function and the return of firm *'i*' in period *t*, while $E(\text{Earnings})_{ijt}$ is the firm *j*'s predicted earnings, given the accounting function of firm *'j*' and the return of firm *'i*' in period *t*. The economic events are held constant using firm *i*'s return in both predictions:

$$\operatorname{Comp}_{ijt} = (-1/16) \times \sum_{t=15}^{t} |E(\operatorname{Earnings}_{iit}) - E(\operatorname{Earnings}_{ijt})|.$$
(5)

The comparability between firm *i*'s and firm *j*'s accounting systems is estimated as the absolute difference between the predicted earnings using firm *i*'s and firm *j*'s accounting functions multiplied by -1. Our comparability measure "Comp_{ijt}" has non-positive values. The comparability is higher between firm "*i*" and firm "*j*" when the comparability measure, i.e. Comp_{ijt}, has higher values because it represents a smaller absolute difference between *E*(Earnings)_{*iit*} and *E*(Earnings)_{*ijj*}. We measure comparability Comp4_{it} as the mean of the four highest comparability scores of firm "*i*" in period "*t*" in the same industry and "CompInd_{it}," which is the median of all of the comparability scores of firm "*i*" in the industry for period "*t*." Moreover, we also use Comp10_{it}, which is the mean of the Comp_{ijt} of the top 10 firms within an industry. These measures of comparability are widely used in accounting research.

4.6 Control variables

We control for several firm-level variables. We control for state ownership because government ownership can influence comparability. As discussed earlier, unlike NSOEs, SOEs have social and political objectives, experience higher principle–principle agency conflict, enjoy preferential treatment and have abundant financing options, which consequently affect the demand for comparability. Therefore, the dummy variable (SOE) for state ownership is used, where 1 indicates that a firm is state-owned and 0 otherwise. We also control for EQ as a measure of information asymmetry and managerial opportunism. We use Dechow et al.'s (1995) model of earnings management to measure EQ. We also replicate our results using alternative measures of EQ, such as those from Dechow and Dichev (2002) and Kothari et al. (2005), and our results remain consistent. Higher values of EQ denote lower EQ and vice versa. We further control for volatility of earnings because when earnings are more volatile, the uncertainty regarding firms is higher, leading to higher information asymmetry, which can result in lower comparability. The volatility is measured as the standard deviation of total sales. We control for size (natural log of total assets) because firms that are larger in size tend to prefer higher availability of information, i.e. lower information asymmetry (Datta et al., 2011). Thus, large firm size is expected to be associated with higher comparability. Moreover, higher scrutiny and political risk faced by larger firms also discipline such firms and can result in higher comparability. Prior studies (Datta et al., 2013) argued that leverage is associated with managerial decisions regarding quality of financial reporting; therefore, we control for leverage (LEV). We further control for return on assets because, when firms report less profits (or loss), pressures from owners and the regulatory pressure of being delisted can influence disclosure choices, such as comparability.

5. Results

5.1 Descriptive statistics and correlation matrices

Table I reports descriptive statistics for all of the variables. In the sample, an average firm comes from an industry with mean value of PMC -0.0195, which is comparable to the PMC values stated by Hoberg and Phillips (2010) and Dhaliwal *et al.* (2014), while the median

decision	n	SD	Minimum	Maximum	Median	Mean	Variables
	10,356	1.6953	-12.9964	-0.0699	-0.3677	-0.8513	Comp4
	10,356	2.0002	-15.0458	-0.1333	-0.5860	-1.1841	Comp10
	10,356	2.5259	-18.9863	-0.8079	-1.8174	-2.5803	CompInd
	10,356	0.03457	-0.2171	00018	-0.0731	01956	PMC
0.454	10,354	0.0916	0.0010	0.5833	0.0546	0.0809	EQ
2451	9,976	0.1479	0.0155	0.8466	0.1433	0.1830	Volatility
	9,978	1.2066	19.4076	25.4741	21.7605	21.8937	Size
	9,791	3.0034	0.6273	19.5021	2.5650	3.4763	MB
	9,978	0.1978	0.0728	0.9413	0.5185	0.5075	Lev
	9,978	0.0605	-0.2014	0.2219	0.0322	0.0366	ROA
	9,889	0.0506	0.2727	0.5714	0.3333	0.3644	Ind_Dir

Notes: Comp4 is the mean of four highest comparability scores of hrm '7' in period '7'; Comp10 is the mean of top 10 firms in an industry; CompInd is the median of all comparability scores of firm 'i'" in the industry for period "t"; EQ represents earnings quality based on discretionary accruals management using modified Jones model as described by Kothari *et al.* (2005), PMC is proxy for product market competition using Herfindahl– Hirschman index (HHI) multiplied by negative one; Size is natural log of total assets; MB is ratio of market to book value of equity; Volatility is standard deviation of sales; Lev represents leverage ratio; ROA represents return on assets; Ind_Dir exhibits board independence, i.e. the ratio of independent directors in the total number of directors on board. This table reports summary statistics for some characteristics of the sample. These statistics are for the period spanning 2005–2014 for the firms meeting data requirement

Table I. Descriptive statistics

value is -0.073. The mean values of the three financial statement comparability measures are -0.851, -1.184 and -2.58, respectively, while the median values of the comparability measures are -0.367, -0.586 and -1.817.

The pairwise correlation between the variables used in the analysis is presented in Table II. The correlations among the control variables, i.e. PMC, EQ, SOE, Size, BM, Volatility, LEV, ROA, Big4 and Ind_Dir, are reported in the table. Competition (PMC) is positively associated with comparability measures, suggesting that firms experiencing intense competition exhibit higher comparability. We also find a negative correlation between absolute discretionary accrual and comparability, suggesting that comparability increases with the increase in accounting quality. The correlation of other control variables and comparability measures also meets expectations.

5.2 Regression analysis

The regression results are reported in Table III for the impact of competition on comparability. The coefficient in model one is 2.464, and it is statistically significant at the conventional level. These results show that, as competition increases, comparability also increases. The coefficients with all three measures of comparability are consistent and statistically significant. These results suggest that competition plays a monitoring role and aligns managerial interests by increasing the quality and transparency of financial reports, thereby reducing information asymmetry. These results suggest that comparability, a qualitative aspect of financial reporting, is influenced by competition. These results suggest that competition improves the quality of financial reporting, resulting in transparent financial reports and lower information asymmetry. These results suggest a new channel through which competition affects financial reporting.

5.2.1 Considering the influences of auditors and independent directors. In model 2, we introduce two more variables to account for the effects of other external governance mechanisms likely to influence the financial reporting and information environment. Two external governance mechanisms used in this study are big 4 auditors and independent directors. Prior research (Choi and Wong, 2007; Wang and Xin, 2011) has reported a



MD 56,11	Ind_Dir	-
	Big4	$1 \\ 0.033**$
2452	ROA	1 0.078*** -0.008
	Lev	1 382**** 0.058****
	MB	1 0.055*** 0.075*** 0.061***
	Size	$\begin{array}{c} 1\\ -0.335^{****}\\ 0.352^{****}\\ 0.154^{****}\\ 0.050^{****}\end{array}$
	Volatility	1 0.004 0.121**** 0.03**** 0.071**** 0.071**** 0.022**
	EQ	$\begin{array}{c} 1\\ 0.193^{***}\\ 0.052^{***}\\ 0.092^{****}\\ 0.067^{****}\\ -0.006\\ -0.032^{****}\\ 0.013\end{array}$
	SOE	1 -0.037*** -0.058*** 0.237*** 0.237*** 0.185*** -0.063*** ectively
	PMC	1 -0.165*** 0.042*** -0.014 -0.552**** 0.106*** -0.109*** -0.03***
	CompInd	1 0.030** 0.004 -0.140*** -0.150*** 0.094*** 0.203*** 0.203*** -0.001 -0.001 -0.001 mind 1 percen
	Comp10	1 0.955*** 0.0100* 0.001 -0.133*** 0.013**** 0.105**** 0.227*** 0.004 -0.006 ant at 10, 5 i
	Comp4	1 0.992**** 0.933*** 0.023*** 0.004 -0.126*** 0.0107**** 0.107**** 0.231**** 0.231****
Table II. Correlation matrix		Comp4 Comp10 Comp10 PMC SOE EQ Volatility Size Lev RDA Big4 Big4 Big4 Big4 Size Notes: *,*
المستشارات		

d Model 2	5.197**** (2.87) 0.451* (1.82) -1.027**** (-3.72) -0.969**** (-3.72) 0.708*** (5.24) -0.148*** (-5.68) -2.582**** (-5.51) 3.524*** (4.39) -0.124 (-0.39) -0.124 (-0.39) -0.124 (-0.39) -15.549**** (-5.49) Yes Yes Yes Yes Yes Yes Yes S.55**** S.55****	Managerial decisions
Complu	5.083**** (2.82) 0.452** (1.87) -1.043**** (-3.76) -1.022**** (-3.94) 0.700**** (5.28) -0.154*** (-5.91) -2.607**** (-5.91) -2.607**** (-5.69) 3.693**** (4.61) -15.478*** (-5.69) Yes Yes 14.66 9.78*** comparability for the sample s include firm and year dun cent levels, respectively	2453
p10 Model 2	3.179** (2.39) 0.341* (1.68) -0.733^{****} (-3.22) -0.903^{****} (-4.01) 0.549^{****} (4.88) -0.115^{****} (-5.13) -0.115^{****} (-5.13) -0.115^{****} (-5.66) 2.550^{****} (3.85) -0.074 (-0.30) -0.381 (-0.62) -0.11234^{****} (-4.79) Yes Yes Yes Yes 15.09 7.13^{****} on on financial statement bility measures. All model mificant at 10, 5 and 1 per-	
Model 1	3.094** (2.34) 0.342* (1.73) -0.753*** (-3.30) -0.948*** (-4.21) 0.546*** (4.95) -0.120*** (-5.36) -2.215*** (-5.22) 2.684*** (4.05) -111.282*** (-5.01) Yes Yes 15.56 8.17*** product market compara ancial statement compara ered by firms. *, **, ****Sig	
mp4 Model 2	2.540** (2.27) 0.292* (1.66) -0.613*** (-3.11) -0.833*** (-4.20) 0.478*** (4.91) -0.098*** (-4.93) -1.878*** (-5.72) -0.0491 (-0.76) -9.557*** (-4.73) Yes Yes Yes 14.67 6.88**** e xamining the effect of dependent variable is fin the standard errors clust	
Model 1	2.464** (2.22) 0.289* (1.69) -0.619^{**} (-3.13) -0.619^{**} (-4.40) 0.476^{***} (4.99) -0.102^{****} (-5.15) -1.898^{****} (-5.83) 2.174^{****} (-5.83) 2.174^{****} (-4.94) Yes Yes Yes 15.16 7.84^{****} sports regression results therefore are calculated w	
	PMC SOE EQ Volatility Size MB MB Lew Big4 Ind_Dir Constant Firm-fixed effects Year-fixed effects Adj. R ² (%) F-statistics Notes: This table r 2005–2014 that meet reported in the pare	Table III. Product market competition and financial statement comparability
الِڭ للاستشارات		W

governance role of big auditors. Similarly, a higher number of independent directors play a governance role and improve firm performance (Duchin *et al.*, 2010). Our results remain robust after the inclusion of both variables. The results in model 2 show that the impact of competition on comparability is robust in the presence of other governance mechanisms. However, the impacts of the big 4 auditors and independent directors on financial statement comparability are statistically insignificant. Prior research has also offered some explanation for this phenomenon. Francis and Wang (2008) reported no difference in the EQ of big 4 and non-big 4 auditors when the institutional environment is weak. China's weak institutional setting provides a rationale for insignificant relationships between the big 4 firms and comparability. Moreover, the big 4 auditing firms also assign less experienced partners to the firms listed in Chinese weak institutional settings (Ke et al., 2015). Similarly, a larger number of independent directors do not improve the comparability in itself, in agreement with prior studies. For example, Wang (2014) reviewed 30 empirical studies exploring the relationship between board independence and firm performance in China, and found that 21 of the 30 studies reported an insignificant relationship.

Our control variables, included in the models reported in Table III, suggest that state ownership, size, EQ and return on assets are positively associated with comparability. The results also show that the volatility of earnings and leverage are negatively associated with comparability. Overall, our control variables meet the expectations.

5.2.2 Industry leaders/followers and comparability. Following Li (2010) and Dhaliwal *et al.* (2014), we divide our sample within the industry into leaders and followers on the basis of the market share of each firm. First, we divide the firms into quartiles on the basis of their market shares, and then those firms in the top quartile are labeled leaders, and those in remaining quartiles are labeled industry followers. We create a dummy variable on the basis of industry leaders/followers, where one represents industry leaders, and zero represents industry followers. According to Dhaliwal *et al.* (2014), "industry leader/follower distinction is useful to discriminate between alternative causal explanations." Our results, as reported in Table IV, meet the expectations. We do not find any statistically significant association between industry leadership and comparability even after controlling for big 4 auditors and board independence (ratio of independent directors).

As an alternative approach, we estimate separate regressions for each sub-group (i.e. leaders/followers). Our results (untabulated) remain similar to those of our main test.

5.2.3 Competition from existing rivals/potential rivals and comparability. Now we examine the effects of competition from existing and potential entrants on comparability. Our results show that competition from existing rivals (PMC_EXST) is positively associated with comparability (Table V).

Although the relationship between competition from potential entrants and comparability is negative, it is statistically insignificant at the conventional level. These results indicate that there is no significant relationship between competition from potential entrants (PMC_POTN) and comparability. These findings strengthen the arguments by Li (2010) that existing competition increases financial disclosure quality. Although these findings are bit different from expectations, they are in line with previous studies (Li, 2010; Majeed and Zhang, 2016) documenting that both dimensions affect corporate decisions in different manners (Table VI).

6. Additional tests and robustness checks

6.1 Competition, state ownership and comparability

The study of SOEs is important because SOEs constitute almost 10 percent of the GDP of the world. SOEs have become more dominant in the world economy recently, particularly



MD

56.11

	Model 2 -0.116 (-0.93) 0.451* (1.82) -0.935*** (-3.74) -0.935*** (-3.74) -0.937*** (-3.85) 0.680**** (-3.72) -0.113** (-5.72) -2.611*** (-5.72) -2.611*** (-5.72) -2.611*** (-5.72) -15.08**** (-5.29) Yes Yes Yes Yes Yes Yes Yes Yes	Managerial decisions
Complia	Model 1 -0.109 (-0.88) 0.452* (1.88) $-1.050^{***} (-3.78)$ $-1.043^{***} (-4.03)$ $0.673^{***} (5.05)$ $-0.155^{***} (-5.95)$ $-2.636^{***} (-5.51)$ $3.648^{***} (-5.48)$ Yes Yes Yes 14.28 9.43^{****} comparability for the sample include firm and year dur recent levels, respectively	2455
01q	Model 2 -0.093 (-0.94) $0.341^{*} (1.68)$ $-0.738^{****} (-3.24)$ $0.916^{***} (-4.08)$ $0.535^{***} (4.75)$ $-0.115^{***} (-5.71)$ $2.528^{****} (-5.71)$ $-2.207^{***} (-5.71)$ -0.076 (-0.31) -0.076 (-0.31) -0.076 (-0.31) $2.528^{***} (-4.67)$ Yes	
Com	Model 1 -0.088 (-0.90) 0.342* (1.73) -0.0558*** (-3.31) -0.0558*** (-4.27) 0.532*** (4.81) -0.121*** (-5.87) 2.661*** (4.02) -2.233*** (-4.88) Yes Yes Yes Ves 15.12 9.95*** f industry leaders/followe ancial statement compara- tered by firms, *,**********************************	
up4	Model z -0.076 (-0.90) 0.292* (1.66) -0.617*** (-3.12) -0.843*** (-4.26) 0.466*** (4.79) -0.098*** (-5.76) -1.893*** (-5.76) -0.051 (-0.24) -0.051 (-0.27) -9.330*** (-4.61) Yes Yes Yes 1.4.35 6.74^{***} 6.74^{***} s examining the effect of dependent variable is fir ith standard errors clust	
Cor	$\begin{array}{c} \text{Model 1} \\ -0.073 (-0.87) \\ 0.289* (1.69) \\ -0.023^{****} (-3.14) \\ -0.880^{****} (-4.45) \\ 0.464^{****} (4.86) \\ -0.103^{****} (-5.16) \\ -1.913^{****} (-5.93) \\ 2.156^{****} (3.69) \\ -1.913^{****} (-5.93) \\ 2.156^{****} (3.69) \\ 1.913^{****} (-5.93) \\ 7.61^{****} \\ eports regression results of the threes are calculated weights of the three threes are calculated weights of the three threes are calculated weights of the three three three threes are calculated weights of the three th$	
	Leaders SOE EQ Volatility Size MB Lev ROA Big4 Ind_Dir Constant Frim-fixed effects Year-fixed effects Adj. R ² (%) F-statistics Notes: This table r 2005–2014 that meet reported in the pare	Table IV. Industry leaders/ followers and financial statement comparability
للاستشارات	isi	w

www

56,11	JInd Model 2	0.197** (2.40) 0.452* (1.83) -1.012**** ($-3.66-0.981$ **** ($-3.820.659$ **** (5.00) -0.149**** ($-5.75-2.583$ ***** ($-5.77-2.583$ ***** ($-5.77-2.583$ ***** ($-5.77-2.583$ ***** ($-5.77-2.14,498$ **** ($-5.72Yes$
2456	Comp Model 1	0.193** (2.36) 0.452* (1.88) -1.028*** (-3.29) 0.653*** (-3.99) 0.653*** (-5.97) -0.155*** (-5.97) -0.155*** (-5.65) 3.645*** (4.55) 3.645*** (4.55) 3.645*** (-5.46) Yes 14.20 9.91**** t comparability for the si sinclude firm and year occut levels, respectively
	10 Model 2	0.140*** (2.17) 0.342** (1.69) -0.721**** (-3.17) -0.908**** (-4.06) 0.519**** (4.73) -0.116**** (-5.19) -0.116**** (-5.19) -0.116**** (-5.19) -0.116**** (-5.19) -0.116*** (-5.19) -0.069 (-0.28) -0.069 (-0.28) -10.52**** (-4.64) Yes 15.38 15.38 15.38 723**** vals on financial statemetion interval on the statemetion of the statemetic of the stat
	Comp Model 1	0.138** (2.14) 0.342** (.1.73) -0.742*** (-3.25) -0.952*** (-4.25) 0.516*** (4.81) -0.121*** (5.41) -0.121*** (5.41) 2.657*** (4.01) 2.657*** (4.01) 2.657**** (4.01) 2.657*** (4.01) 2.657**** (4.01) 2.657*** (4.01) 2.657****
	p4 Model 2	0.119** (2.13) 0.293* (1.67) -0.603*** (-3.06) -0.837*** (-4.25) 0.453*** (4.78) -0.099*** (-4.98) -1.876*** (-5.71) 2.039*** (-4.98) -0.045 (-0.21) -0.045 (-0.21) -0.046 (-0.75) -0.046 (-0.75) -0.047 (-0.75) -0.040
	Com Model 1	0.116** (2.10) 0.290* (1.70) -0.610**** (-3.08) -0.875**** (-4.44) 0.451**** (-5.19) -0.103**** (-5.19) -1.896**** (-5.19) -1.896**** (-4.83) Yes Yes Yes 15.35 T2.3**** (-4.83) Yes 15.35 T2.3**** (-4.83) Yes 15.35
Cable V. competition from xisting rivals and nancial statement comparability		PMC_EXST SOE EQ Volatility Size MB MB Big4 Ind_Dir Constant Firm-fixed effects Year-fixed effects Adj. R ² (%) <i>F</i> -statistics Notes: This table rep <i>P</i> -statistics Notes: This table rep

WWW

I Model 2	$\begin{array}{c} -0.056 \ (-0.32) \\ 0.274 \ (1.15) \\ 0.274 \ (1.15) \\ 0.275 \ (-3.72) \\ 0.595 \ (-4.28) \\ 0.595 \ (-4.28) \\ 0.595 \ (-4.28) \\ 0.295 \ (-4.28) \\ 2.038 \ (-4.56) \\ 2.038 \ (-4.56) \\ 2.038 \ (-4.56) \\ 2.038 \ (-4.56) \\ 2.038 \ (-4.56) \\ 1.013 \\ -0.766 \ (-1.01) \\ -1.031 \\ 2.138 \ (-4.56) \\ 1.031 \\ 0.062 \ (0.18) \\ -0.768 \ (-1.01) \\ 1.031 \\ 0.062 \ (0.18) \\ 1.031 \\ 0.014 \ (-4.56) \\ 1.003 \ (-4.56) \ (-4.5$	Managerial decisions
Complnc Model 1	$\begin{array}{c} -0.055 \ (-0.31) \\ 0.292 \ (1.27) \\ 0.2966*** \ (-3.56) \\ -0.801 \ *** \ (-3.14) \\ 0.595*** \ (4.33) \\ 0.595*** \ (4.33) \\ 0.595*** \ (-3.14) \\ 0.595*** \ (-4.78) \\ -1.32*** \ (-4.70) \\ 2.226*** \ (2.61) \\ 2.226*** \ (2.61) \\ 2.226*** \ (2.61) \\ 2.226*** \ (2.61) \\ 1.0.97 \\ 7.60*** \\ 3.01 \ 10.97 \\$	2457
p10 Model 2	$\begin{array}{c} -0.097 \ (-0.85) \\ 0.150 \ (0.77) \\ -0.687^{****} \ (-3.12) \\ -0.746^{****} \ (-3.12) \\ -0.7482^{****} \ (-4.15) \\ -0.482^{****} \ (-4.15) \\ -1.854^{****} \ (-4.15) \\ 1.763^{****} \ (-4.14) \\ 1.763^{****} \ (-4.18) \\ -0.769 \ (-1.29) \\ -0.769 \ (-1.29) \\ -0.769 \ (-1.29) \\ -0.769 \ (-1.29) \\ -0.769 \ (-1.29) \\ -0.769 \ (-1.29) \\ 1.49 \\ \text{Yes} \\ Yes \\ Yes \\ 1.49 \\ \text{ial entrants on financial i i to, 5i \\ 5.60^{****} \\ \text{is ****Significant at 10, 5i \\ *****Significant at 10, 5i \\ \text{is ****Significant at 10, 5i \\ \text{is *****Significant at 10, 5i \\ \text{is *****} \\ \text{is *****Significant at 10, 5i \\ \text{is ****} \\ \text{is *****} \\ \text{is *****} \\ \text{is *****} \\ \text{is ****} \\ \text{is ***} \\ \text{is ****} \\ \text{is ***} \\ \text{is **} \\ \text{is ***} \\ \text{is ***}$	
Com Model 1	$\begin{array}{c} -0.097 \ (-0.85) \\ 0.172 \ (0.91) \\ -0.0574^{****} \ (-3.02) \\ 0.785^{****} \ (-3.29) \\ 0.484^{****} \ (4.29) \\ 0.086^{****} \ (-4.42) \\ 1.886^{****} \ (-4.42) \\ 1.886^{****} \ (-4.42) \\ Yes \\ Yes \\ Yes \\ Yes \\ Yes \\ Yes \\ 12.15 \\ 6.40^{****} \\ e.40^{****} \\ e.tompetition from potent atterment able is financial statement able is financial statement or scuttered by firms. *, *$	
np4 Model 2	$\begin{array}{c} -0.067 \ (-0.71) \\ 0.113 \ (0.69) \\ 0.113 \ (0.69) \\ 0.133 \ (0.69) \\ 0.0131 \ (0.68) \\ 0.01414^{***} \ (4.25) \\ 0.01414^{***} \ (4.25) \\ -0.0768^{***} \ (-3.97) \\ -1.573^{***} \ (-4.90) \\ 1.548^{**} \ (2.57) \\ 0.003 \ (0.36) \\ 0.033 \ (0.36) \\ -0.681 \ (-1.30) \\ -0.681 \ (-1.30) \\ -0.681 \ (-1.4) \\ Yes $	
Con Model 1	$\begin{array}{c} -0.069 \ (-0.72) \\ 0.129 \ (0.81) \\ 0.129 \ (0.81) \\ 0.417^{***} \ (-2.85) \\ 0.417^{***} \ (-3.33) \\ 0.417^{***} \ (-3.33) \\ 0.417^{***} \ (-4.38) \\ 1.680^{***} \ (-4.38) \\ 1.680^{***} \ (-4.38) \\ Yes \\ Yes \\ Yes \\ Yes \\ Yes \\ 12.12 \\ 5.74^{***} \\ eports regression results that meet data requirement the parentheses are calcu$	
	PMC_POTN SOE EQ Volatility Size MB Lev ROA Big4 Ind_Dir Constant Firm-fixed effects Year-fixed effects Adj. R ² (%) F-statistics Notes: This table r spanning 2005–2014 t-values reported in	Table VI. Competition from potential entrants and financial statement comparability
غ للاستشارات	isi	w

after the financial crisis of 2008. SOEs are not only important in developing economies but are also important in developed countries. SOEs constitute almost 5 percent of the GDP of OECD member countries (Peng *et al.*, 2016). Although the Chinese corporate sector is comprised of a large number of SOEs (Wu *et al.*, 2016), the study of SOEs is important and relevant for all developed and developing economies.

These SOEs are different from NSOEs in terms of objectives, financial reporting practices, riskiness, performance and governance mechanisms (Allen et al., 2012). An institutional environment with large number of SOEs, such as China, not only suffers from agency conflicts between managers and shareholders (Jensen and Meckling, 1976) but also between majority and minority shareholders, i.e. principle-principle agency conflicts (Ali et al., 2007). Khanna and Rivkin (2001) also argued that concentrated ownership and weak investor protection permit the controlling shareholders to expropriate minority investors. Moreover, profit maximization is not the primary objective of SOEs in China, but they also pursue other social and political goals set by the government (Faccio, 2006). SOEs are viewed as less risky because, in cases of distress, they can obtain financial support from the state (Chen *et al.*, 2010). The big 4 banks, owned by the government, provide almost 80 percent of the total industrial and commercial loans. SOEs also receive preferential treatment from these state-owned banks. These arguments suggest that SOEs face fewer capital market pressures because they are not dependent on capital markets for funds. Prior studies (e.g. Shen and Lin, 2009) have documented that informal networks and government ownership are key governance mechanisms in China. Such differences also cause the financial reporting of SOEs to differ from that of NSOEs.

This study exploits such differences in institutional environments and examines how state ownership affects the relationships between competition and comparability. We examine how the relationship between competition from existing and potential rivals and comparability is affected by state ownership. Tables VII and VIII report the results of this effect of state ownership on the relationship between competition and comparability. The interaction term EXST_SOE (POTN_SOE) shows the influence of state ownership on the relationship between competition from existing rivals (potential entrants) and comparability.

The main variables EXST_SOE and POTN_SOE are statistically insignificant. These results suggest that state ownership has no impact on the relationship between competition and comparability, i.e. competition plays no role in increasing (decreasing) comparability for SOEs, consistent with Majeed and Zhang (2016), who argued that competitive pressure plays no role in shaping the EQ of SOEs. These findings also suggest that competition might not mitigate the agency conflict between majority and minority shareholders (agency conflict). These findings are also in line with Cormier *et al.* (2016), who argued that the usefulness of a governance mechanism is context-dependent, and governance mechanism can require complementary mechanisms under certain circumstances.

6.2 NPC and comparability

We also examine the impact of NPC on comparability. We use the industry-level advertising-to-sales ratio as a measure of NPC following Chen *et al.* (2015). Advertising has been characterized as "the prototype of non-price variables" (Stigler, 1968). The preceding economic research treats reduced prices and higher advertising expenses as price and non-price approaches, respectively, to escalate demand for any product or service. Schmalensee (1978) explained how advertising could prevent entrance into NPC industries (e.g. ready-to-eat breakfast cereal): "the products in this market are clearly differentiated, and advertising-to-sales ratio has been used as a measure of NPC in the recent literature (e.g. Chen *et al.*, 2015). Table IX reports the results for NPC and comparability, suggesting a positive association between NPC and comparability.



MD

56.11

Model 2	0.277^{***}_{**} (2.59) 0.486^{**} (1.94) -0.142 (-1.45) -0.142 (-1.45) -0.138 (-3.65) -0.981^{***} (-3.65) 0.656^{***} (4.99) -0.150^{***} (-5.45) 3.471^{***} (-5.45) 3.471^{***} (-5.45) -0.275 (-0.36) -0.275 (-0.36) -0.275 (-0.36) Yes Yes 13.855 8.31^{***} ensure as defined resure as defined resure as defined resure as defined	Managerial decisions
Complné Model 1	$\begin{array}{c} 0.271^{**} \ (2.55) \\ 0.484^{*} \ (1.99) \\ -0.138 \ (-1.419) \\ -0.138 \ (-7.410) \\ 0.1026^{***} \ (-3.69) \\ -1.026^{***} \ (-3.69) \\ 0.0550^{***} \ (-5.60) \\ 0.0550^{***} \ (-5.99) \\ -1.0449^{***} \ (-5.99) \\ -2.567^{***} \ (-5.99) \\ 3.641^{***} \ (+.54) \\ -14.49^{***} \ (-5.46) \\ Yes \\ Yes \\ 14.35 \\ 9.39^{***} \ (-5.46) \\ Yes \\ 14.35 \\ Yes \\ Yes \\ 14.35 \\ Yes \\ Yes \\ 14.35 \\ Yes \\ 14.35 \\ Yes \\ Yes \\ 14.35 \\ Yes \\ Yes \\ Yes \\ 14.35 \\ Yes \\ Y$	2459
p10 Model 2	$\begin{array}{c} 0.196^{**} \ (2.28) \\ 0.365^{*} \ (1.78) \\ -0.038 \ (-1.28) \\ -0.038 \ (-1.28) \\ -0.908^{***} \ (-3.16) \\ 0.516^{****} \ (-3.16) \\ 0.516^{****} \ (-5.19) \\ -0.116^{****} \ (-5.19) \\ -2.157^{****} \ (-5.61) \\ 2.2520^{****} \ (-5.61) \\ -2.157^{***} \ (-4.64) \\ Yes \\ Yes \\ 15.50 \\ 6.94^{****} \\ (-4.64) \\ Yes \\ 15.50 \\ 6.94^{****} \end{array}$	
Comp Model 1	0.191** (2.24) 0.364* (1.82) -0.094 (-1.23) -0.740*** (-3.24) 0.514*** (-3.24) 0.514*** (-4.26) 0.514*** (-4.26) 2.654*** (4.00) -0.121*** (-5.78) 2.654*** (4.00) 2.654*** (-4.86) Yes 15.98 7es 15.98 7es 15.98 7es 15.98 7es 15.98 7es 15.98 7es 16.988 7es 16.988 7es 16.988 7es 16.988 7es 16.988 7es 16.988	
mp4 Model 2	0.165** (2.22) 0.313* (1.76) -0.083 (-1.25) -0.083 (-1.25) -0.602^{***} (-3.05) -0.837^{***} (-4.26) 0.451^{****} (-4.26) -0.094^{***} (-4.98) -0.04^{*} (-0.74) -0.04^{*} (-0.74) -0.04^{*} (-0.74) -0.19) -0.04^{*} (-0.74) -0.74^{*} Yes Yes Yes 14.95 6.70^{****} 14.95 6.70^{****} 14.95 6.70^{****} 14.95 14.95 32005-2014 that meet dat mies and t -values report	
Con Model 1	0.161** (2.18) 0.308* (1.78) -0.08 (-1.21) -0.08 (-1.21) 0.450**** (-3.07) 0.450**** (-4.44) 0.450**** (-5.20) -1.872**** (-5.84) 2.150**** (-4.83) Yes 1.872**** (-4.83) Yes 1.872**** (-4.83) Yes 1.5.45 7.57**** reson results, resonable firms spanning ports regression results, resonable firm and year dun levels, respectively	
	PMC_EXST SOE EXST_SOE EQ Volatility Size MB Size MB Big4 Ind_Dir Constant Firm-fixed effects Adi, R^2 (%) <i>F</i> -statistics Adi, R^2 (%) <i>F</i> -statistics Notes: This table r comparability for th earlier. All models i 10, 5 and 1 percent	Table VII. State ownership, competition from existing rivals and financial statement comparability
للاستشارات		W

MD 56,11	Ind Model 2	$\begin{array}{c} -0.528 \ (-0.80) \\ 0.346 \ (1.39) \\ 0.346 \ (1.39) \\ 0.346 \ (1.39) \\ 0.346 \ (1.39) \\ 0.060 \ (1.8) \\ -0.746 \ (-4.57) \\ 0.060 \ (0.18) \\ -0.125 \ (-4.51) \\ -2.149 \ (-4.57) \\ 2.083 \ (-4.57) \\ 2.083 \ (-4.57) \\ 2.083 \ (-4.57) \\ 2.083 \ (-4.59) \\ -0.773 \ (-1.02) \\ -1.339 \ (-4.59) \\ Yes \\ Yes \\ Yes \\ Yes \\ 10.27 \\ 6.40 \ (-4.59) \\ Yes \\ Arrants and financial arability measure as clustered by firms. \end{array}$
2460	Comp Model 1	$\begin{array}{c} -0.539\ (-0.82)\\ 0.367\ (1.52)\\ 0.367\ (1.52)\\ 0.506\ (0.78)\\ -0.797^{***}\ (-3.12)\\ 0.600^{****}\ (+.3.12)\\ 0.600^{****}\ (+.4.35)\\ -0.131^{****}\ (-5.03)\\ -2.200^{****}\ (-4.80)\\ Yes\\ 10.92\\ Yes\\ 10.92\\ 7.18^{****}\\ qpetition from potential et s financial statement complexity is the statement of the st$
	p10 Model 2	$\begin{array}{c} -0.325 \ (-0.67) \\ 0.185 \ (0.92) \\ 0.238 \ (0.51) \\ -0.687^{***} \ (-3.12) \\ -0.745^{***} \ (-3.12) \\ -0.745^{***} \ (-4.15) \\ -0.091^{***} \ (-4.15) \\ -1.860^{***} \ (-4.15) \\ 1.775^{***} \ (2.57) \\ 0.078 \ (0.29) \\ -0.773 \ (-1.29) \ (-1.29) \\ -0.773 \ (-1.29) \ (-1.29) \\ -0.773 \ (-1.29) \ (-1.29) \ (-1.29) \\ -0.773 \ (-1.29) \ (-1.$
	Com Model 1	$\begin{array}{c} -0.332 \ (-0.68) \\ 0.208 \ (1.06) \\ 0.208 \ (1.06) \\ 0.245 \ (0.52) \\ -0.674^{***} \ (-3.02) \\ -0.783^{***} \ (-3.37) \\ 0.487^{***} \ (-4.37) \\ 0.487^{***} \ (-4.43) \\ 1.897^{***} \ (-4.43) \\ 1.897^{***} \ (-4.43) \\ Yes \\ Yes \\ Yes \\ 12.14 \\ 6.05^{***} \\ f \ state \ ownership \ on \ the \\ t \ meet \ data \ requirements. \end{array}$
	np4 Model 2	$\begin{array}{c} -0.372 \ (-0.89) \\ 0.16 \ (0.93) \\ 0.16 \ (0.93) \\ 0.318 \ (0.80) \\ -0.566^{****} \ (-2.98) \\ -0.698^{****} \ (-3.67) \\ 0.417^{****} \ (-2.98) \\ 0.698^{****} \ (-3.97) \\ -1.569^{****} \ (-3.97) \\ -1.564^{****} \ (-3.97) \\ -1.564^{****} \ (-3.97) \\ -1.564^{****} \ (-3.97) \\ -2.422^{****} \ (-4.17) \\ Yes \\ Yes \\ Yes \\ Yes \\ 11.41 \\ 4.79^{****} \\ s \ examining the effect of that and the other of the other of the other of the other other$
	Cor Model 1	$\begin{array}{c} -0.382 \ (-0.91) \\ 0.178 \ (1.06) \\ 0.327 \ (0.82) \\ -0.548^{***} \ (-2.85) \\ -0.548^{****} \ (-2.85) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 0.327 \ (0.82) \\ 1.695^{***} \ (-4.42) \\ Yes \ Yes \ Yes \ Yes \ Yes \ Yes \ (-4.42) \\ Yes \ Yes \ (-4.42) \\ Yes \ (-4.42) \\ Yes \ (-4.42) \\ 10.95^{***} \ (0.6110 \ (0.6$
Table VIII. State ownership, competition from potential entrants and financial statement comparability		PMC_POTN SOE POTN_SOE EQ Volatility Size MB ROA Big4 Lev ROA Big4 Ind_Dir Constant Firm-fixed effects Year-fixed effects Year-fixed effects This table r statement comparable defined earlier. All *,******Significant a
الم للاستشارات	i	

odel 1 Model 2	$\begin{array}{cccccc} 7^{***} \left(2.23 \right) & 0.018^{***} \left(2.36 \right) \\ 3.33 \left(1.43 \right) & 0.329 \left(1.37 \right) \\ 8^{***} \left(-3.57 \right) & 0.928^{****} \left(-3.57 \right) \\ 8^{***} \left(-3.57 \right) & -0.928^{****} \left(-3.08 \right) \\ 8^{****} \left(-3.08 \right) & 0.470^{****} \left(-3.08 \right) \\ 8^{****} \left(-6.47 \right) & -0.159^{****} \left(-6.17 \right) \\ 8^{****} \left(-6.47 \right) & -0.159^{****} \left(-6.17 \right) \\ 8^{****} \left(-4.55 \right) & -0.2077^{****} \left(-6.17 \right) \\ 8^{****} \left(3.48 \right) & -0.110 \left(-0.31 \right) \\ 8^{****} \left(3.48 \right) & -0.110 \left(-0.31 \right) \end{array}$	-0.590 (-0.77 Yes Yes Yes Yes Yes Yes Yes Yes 10,680*** - 3.96 Yes Yes Yes 14,05 13*** 7.09*** ar dumnies and t-values reported ir ely	Managerial decisions 2461
.10 Model 2 M	$\begin{array}{c} 0.016^{*} (1.84) & 0.017 \\ 0.245 (1.25) & 0.236 \\ -0.681^{***} (-3.19) & -0.936 \\ -0.688^{***} (-3.48) & -0.784 \\ 0.356^{***} (-3.48) & -0.784 \\ 0.356^{***} (-3.44) & -0.784 \\ -0.123^{***} (-5.40) & -0.167 \\ -1.787^{***} (-4.59) & -2.1147 \\ 1.891^{***} (2.72) & 2.986 \\ -0.058 (-0.21) & 2.0867 \\ \end{array}$	-0.673 (-1.13) -7.099*** -3.31 Yes Yes 14.79 5.43*** 8	
Model 1	$\begin{array}{c} 0.018^{***} (2.00) \\ 0.249 (1.30) \\ -0.692^{****} (-3.20) \\ -0.739^{****} (-3.26) \\ 0.350^{****} (-3.45) \\ 0.350^{****} (-3.45) \\ -0.129^{****} (-5.68) \\ -1.823^{****} (-4.75) \\ -1.823^{****} (-4.75) \\ 2.117^{****} (3.02) \end{array}$	-7.311**** -3.53 Yes Yes 15.29 6.38*** 6.38*** for compartion on finan eptice compartion on finan eptice compartion on finan eptice compartion of finan	
np4 Model 2	$\begin{array}{c} 0.021 ^{***} (2.68) \\ 0.216 (1.28) \\ -0.574 ^{***} (-3.16) \\ -0.640 ^{***} (-3.69) \\ 0.032 ^{***} (3.45) \\ -0.105 ^{***} (-4.67) \\ 1.515 ^{***} (-4.67) \\ 1.515 ^{***} (2.48) \\ -0.035 (-0.15) \end{array}$	-0.663 (-1.27) -5.912*** -3.29 Yes Yes 14.46 5.06*** amining the effect of non- ariable is financial statem errors clustered by firms.	
Model 1	$\begin{array}{c} 0.022^{****} & (2.83) \\ 0.215 & (1.32) \\ -0.572^{****} & (-3.11) \\ -0.685^{****} & (-3.38) \\ 0.302^{****} & (-3.48) \\ 0.302^{****} & (-4.43) \\ -1.555^{****} & (-4.43) \\ 1.706^{****} & (2.77) \end{array}$	-6.130*** -3.53 Yes Yes 15.08 5.83*** 5.83*** corts regression results ex ements. The dependent v alculated with standard.	
	PMC_NPC SOE EQ Volatility Size MB Lev ROA Big4	Ind_Dir Constant Firm-fixed effects Year-fixed effects Adj. R^2 (%) F-statistics Notes: This table rep that meet data requir the parentheses are o	Table IX. Non-price competition and financial statement comparability
فلاستشارات	Lis		w

MDThese results remain consistent after the inclusion of other governance mechanisms (i.e. big56,114 auditors and board independence) and for all measures of comparability. Although
Chen *et al.* (2015) also used the research and development to sales ratio as a second measure
of NPC, R&D disclosures are incomplete and not available for the whole sample period.
However, when we use R&D to sales, our results (untabulated) remain consistent, but
following the approach used by Chen *et al.* (2015), i.e. replacing missing values with zero,
would result in a much greater potential downward bias because of a large number
of missing values. Hence, we report our results with the advertising-to-sales ratio as a
measure of NPC.

6.3 Robustness tests

6.3.1 Alternative measures of comparability. Following prior studies (Chircop *et al.*, 2016; Campbell and Yeung, 2016), we use alternative measures of comparability. Comparability (*ComAcct_{Ajt}*) means estimating firm A's earnings under the assumption that the same business shock is experienced by firm A as experienced by peer firm *R*. For this purpose, underlying business shocks are kept constant for both firms. Stock returns are used as a proxy for business shocks following De Franco *et al.* (2011). We assume that the relationship between earnings and returns is not the same when the firms experience losses, compared to the years when they mad profits (i.e. "the earnings-return relation is asymmetric with respect to whether firms experience positive or negative stock returns") (Chircop *et al.*, 2016). For this purpose, we adapt Basu's (1997) model, and a dummy variable (*D*) is included for negative returns. Furthermore, an interaction term between *D* and "Return" is also introduced. Thus, the influence that business shocks have on firm *A*'s and firm *R*'s earnings is estimated by coefficients as shown in following equations:

Earnings_{*Aig*} = $\alpha_i + \beta_i \text{Return}_{Ajq} + \gamma_i D_{Ajq} + \lambda_i D_{Ajq} \times \text{Return}_{Ajq} + \varepsilon_{Ajq}$,

Earnings_{*Rig*} = $\alpha_R + \beta_i$ Return_{*Rig*} + $\gamma_i D_{Rig} + \lambda_i D_{Rig} \times \text{Return}_{Rig} + \varepsilon_{Rig}$,

where D is one if return is negative and zero otherwise, and the remainder of the procedure is the same as discussed previously in Section 4.3. First, we calculate conditional earnings, and then we estimate the mean difference between quarterly conditional earnings, which are absolute values. These values are then multiplied by negative one, so the higher values represent higher comparability and vice versa. Panel A in Table X reports the results for competition and comparability for the alternative measure of comparability.

Our results are consistent with the prior results reported in Tables III, V and VI. Our results indicate that competition increases comparability. Moreover, competition from existing rivals is also positively associated with comparability, but there is no statistically significant association between competition from potential rivals and comparability. We omit the control variables for brevity.

6.3.2 Taking lag of competition measures. Following prior research (Aschhoff and Schmidt, 2008; Stiebale, 2011; Clemens *et al.*, 2012), we take the first lag of all competition measures and re-estimate our results. All of the results remain consistent. Panel B in Table X reports the results for the first lag of product market competition measures and comparability. We omit the control variables for brevity. Our results remain consistent when we take the second lag to examine the relationship.

6.3.3 Two-stage least square (2SLS) results. Panel C in Table X reports the results for the 2SLS model. For the 2SLS model, predicted values of PMC/PMC_EXST/PMC_POTN are calculated. For this purpose, in first-stage regression, our dependent variables are PMC/PMC_EXST/PMC_POTN, and the independent variables include industry averages of



^{7 cs} ^{9 229} ^{9 229}	00.6	9.29 5 00***	PMC_POTN -0.112 (-0.56) -13.840*** (-4.89)	11.20 3.25^{***}	PMC_POTN 0.301 (0.33) -70.920**** (-3.06) Yes	
Yes 11.19 6.81**** 6.81**** 0.038** (-2.54) 0.761 (-1.25) Yes 24.71 50.99***	10.0	11.19 681***	CompInd PMC_EXST 0.290*** (2.96) -16.734*** (-5.34)	17.70 4.88***	Complud PMC_EXST 0.895* (1.78) -75.778*** (-3.50) Yes	
Yes 11.80 6.84*** 4.524** (2.08) Yes 2.435** (-2.42) Yes 2.4.72 52.28***	0.04	11.80 6.84***	PMC 8.803*** (4.24) -18.516*** (-5.78)	18.40 4.34^{***}	PMC 17.906** (2.16) -79.705*** (-3.57) Yes	
Yes 10.32 4.42*** 0.285 (-1.21) -0.763 (-1.17) Yes 23.59 39.80***		10.32 4.42***	PMC_POTN -0.15 (-1.24) -10.309**** (-4.45)	11.30 2.65***	PMC_POTN 0.255 (0.28) -68.255*** (-2.97) Yes	
Yes 13.07 6.39*** 6.39*** (-13.95) 0.307 (-0.62) Yes 24.80 50.53***	60.0	13.07 6.39***	Comp10 PMC_EXST 0.218*** (2.75) -12.449*** (-4.87)	17.50 3.75^{***}	Comp10 PMC_EXST 0.844* (1.69) -73.207**** (-3.40) Yes	
Yes 12.95 6.42*** 6.42*** (-10.91) PMC 19.158*** (-10.91) -6.538*** (-8.07) Yes 24.40 46.31***	0.42	12.95 642***	PMC 6.390*** (3.92) -13.751*** (-5.26)	17.70 3.32^{***}	PMC 16.078** (1.97) -76.792*** (-3.450 Yes	
Yes 10.60 4.12*** -0.208 (-1.16) Yes 22.91 33.84*** evels, respectivel		10.60 $4.12***$	PMC_POTN -0.116 (-1.15) -8.606*** (-4.32)	10.90 2.37***	PMC_POTN 0.278 (0.31) -66.863**** (-2.92) Yes	
Yes 12.91 6.27*** 6.27*** Comp4 PMC_EXIST 0.171*** (-10.24) 0.084 (-0.20) Yes 23.66 39.40*** i and 1 percent le	17.0	12.91 6 <i>27</i> ***	Comp4 PMC_EXST 0.189*** (2.74) -10.540*** (-4.82)	16.30 3.54^{***}	Comp4 PMC_EXST 0.828* (1.67) -71.886*** (-3.36) Yes	
Yes 12.85 6.31**** 6.31**** (-7.87) -4.151*** (-7.87) -4.151*** (-7.87) -4.151*** (-5.96) Yes 23.43 36.98**** genificant at 10, 5	TC.0	12.85 6.31***	PMC 5.245*** (3.77) -11.621*** (-5.20)	16.50 3.12^{***}	PMC 15.740** (1.95) -75.398*** (-3.41) Yes	
Firmuyear-inxed effects Adj. R^2 (%) Fatistics PMC Statistics Adj. R^2 (%) Fatistics Notes: ******Si	r-stausucs	effects Adj. R^2 (%) F-statistics	Panel B PMC Constant	effects Adj. R^2 (%) <i>F</i> -statistics	Panel A PMC Constant Firm/year-fixed	
W			المن		لف للاسا	شارات

www.

MD competition measures, along with all of the control variables. Then, the predicted values of PMC/PMC_EXST/PMC_POTN are used in the second-stage regression. Our results are consistent and suggest that competition increases comparability. We omit the control variables for brevity.

7. Discussion and conclusions

2464

The objective of this study was to examine the effect of competition on financial statement comparability. We studied whether competition is a significant determinant of comparable financial statements. Our findings reveal that competitive pressure plays a disciplinary role and reduces information asymmetry, leading to higher comparability. We divided the competition into two distinct dimensions, i.e. competition from existing rivals and competition from potential entrants, and we studied their roles in the production of comparable financial statements. The findings suggest that competition from existing rivals results in higher comparability, but competition from potential rivals plays no statistically significant role in improving the comparability of financial statements. Moreover, we also studied the role that the industry leadership plays in shaping comparability, and we report that industry leaders do not face stiff competitive pressure and have fewer comparable financial statements, compared to industry followers.

Our study provides support for the argument that competitive pressure acts as an external governance mechanism, aligns managerial interests and reduces agency conflicts. When agency slack decreases, the incentives for an opaque information environment decrease, and in turn, comparability increases. These findings provide empirical evidence for prior theoretical models (Alchian, 1950; Stigler, 1958; Hart, 1983; Nalebuff and Stiglitzs, 1983; Scharfstein, 1988) that suggested a disciplinary role for competition. In particular, our results are in line with Chen et al. (2014) and Majeed and Zhang (2016), who advocated for a governance role played by competition in China. Nevertheless, this study contests the proprietary cost hypothesis (Verrecchia, 1983; Ali et al., 2014), i.e. competition enhances proprietary costs, resulting in an opaque information environment. Moreover, our study does not support the "dark side" of competition view either, i.e. competition intensifies managerial opportunism (e.g. Lee and Liu, 2016; Lin et al., 2015). Our findings reinforce the argument that the effect of competitive pressure is different on industry leaders and industry followers, which translates into the financial reporting of firms. This study also deepens our understanding regarding the effects of various dimensions of competition (e.g. competition from existing/potential rivals) on the qualitative aspects of financial reporting (i.e. comparability).

In our additional analysis, we examined the role of another important aspect of competition, i.e. NPC, in comparability. Our results reveal that NPC also significantly increases comparability. Bearing in mind the presence of large numbers of SOEs in China and the economic contributions of SOEs globally, we studied the impact of state ownership on the relationship between competition and comparability. The findings of the study suggest that competitive pressure plays no statistically significant role in improving the comparability of SEOs. Because our results suggest that competitive pressure plays no significant role in enhancing the comparability of SOEs, we argue that competitive pressure does not mitigate the agency problem between majority (the state in this case) and minority shareholders (i.e. principle–principle agency conflict) in China, in agreement with Majeed and Zhang (2016), who argued that competitive pressure plays no significant role in improving the EQ in SOEs. However, these findings contradict the argument posed by He (2012), who argued that competitive pressure actually curbs the principle–principle agency conflict. Hence, the role played by competitive pressure, mitigating the agency conflict between majority and minority shareholders, is quite different in Chinese settings.



Therefore, the effect of competitive pressure on principle–principle agency conflicts might be context (institutional settings) dependent.

This study has numerous policy implications. First, our study suggests that, while examining the disclosure practices, importance should be given to the competitive position of the firm and/or level of competition in an industry. Our study also contains some policy implications for standard setters, as they should emphasize non-competitive industries more, with potentially different sets of rules and regulations (to make financial statements more comparable) to decrease information asymmetry. Our study also assists investors in comprehending competition and comparability and could improve their decisions. As competition increases the comparability of financial reports, the investors might be able to make more informed decisions based on knowledge of competitive positions of firms. Investment professionals, rating agencies and analysts should place importance on the competitive position of the firm to assess their information environment and agency problems. As comparability reduces expected stock price crash risks (Kim *et al.*, 2016), the corporate sector could realize the benefits of comparability and the prices that it would have to pay for lower comparability, which could result in firms pursuing higher comparability. The results also have some implications for regulators, as they could improve the overall corporate information environment by increasing competition.

However, there are certain limitations to this study as well. The unique institutional environment of China is quite different from that of other developed and developing economies. Hence, the findings based on the Chinese environment cannot be generalized. Although this study considers it a limitation, this fact might actually deepen our understanding of the contextual influences on this relationship and inspires future studies focusing on other contexts. Research on other economies with different institutional backgrounds might enhance our understanding of the nexus of competition and comparability.

References

- Akdoğu, E. and MacKay, P. (2009), "Investment and competition", Journal of Financial and Quantitative Analysis, Vol. 43 No. 2, pp. 299-320.
- Akdoğu, E. and MacKay, P. (2012), "Product markets and corporate investment: theory and evidence", Journal of Banking & Finance, Vol. 36 No. 2, pp. 439-453.
- Alchian, A.A. (1950), "Uncertainty, evolution, and economic theory", Journal of Political Economylitical Economy, Vol. 58 No. 3, pp. 211-221.
- Ali, A., Chen, T. and Radhakrishnan, S. (2007), "Corporate disclosures by family firms", *Journal of Financial Economics*, Vol. 44 Nos 1-2, pp. 238-286.
- Ali, A., Klasa, S. and Yeung, P.E. (2014), "Industry concentration and corporate disclosure policy", *Journal of Accounting and Economics*, Vol. 58 Nos 2-3, pp. 240-264.
- Alimov, A. (2014), "Product market competition and the value of corporate cash: evidence from trade liberalization", *Journal of Corporate Finance*, Vol. 25 No. 1, pp. 122-139.
- Allen, F., Qian, J., Zhang, C. and Zhao, M. (2012), "China's financial system: opportunities and challenges", in Fan, J. and Morck, R. (Eds), *Capitalizing China*, University of Chicago Press, Chicago, IL, pp. 63-143.
- Aschhoff, B. and Schmidt, T. (2008), "Empirical evidence on the success of R&D cooperation happy together?", *Review of Industrial Organization*, Vol. 33 No. 1, pp. 41-62.
- Baggs, J. and Bettignies, J. De (2007), "Product market competition and agency costs", *The Journal of Industrial Economics*, Vol. LV No. 2, pp. 289-323.
- Balakrishnan, K. and Cohen, D.A. (2014), "Competition and financial accounting misreporting", SSRN Working Paper No. 2447882, September.



Managerial decisions

MD 56.11	Ball, R., Robin, A. and Wu, J.S. (2003), "Incentives versus standards: properties of accounting income in four East Asian countries", <i>Journal of Accounting and Economics</i> , Vol. 36 Nos 1-3, pp. 235-270.
00,11	Barth, M.E. (2013), "Global comparability in financial reporting: what, why, how, and when?", <i>China Journal of Accounting Studies</i> , Vol. 1 No. 1, pp. 2-12.
	Barth, M.E., Landsman, W.R., Lang, M. and Williams, C. (2012), "Are IFRS-based and US GAAP-based accounting amounts comparable?", <i>Journal of Accounting and Economics</i> , Vol. 54 No. 1, pp. 68-93.
2466	Basu, S. (1997), "The conservatism principle and the asymmetric timeliness of earnings", <i>Journal of Accounting & Economics</i> , Vol. 30 No. 1, pp. 3-37.
	Byoun, S. and Xu, Z. (2016), "Product market competition and financial decisions during a financial crisis", <i>Financial Management</i> , Vol. 45 No. 2, pp. 267-290.
	Cai, H. and Liu, Q. (2009), "Competition and corporate tax avoidance: evidence from Chinese industrial firms", <i>The Economic Journal</i> , Vol. 119 No. 537, pp. 764-795.
	Campbell, J.L. and Yeung, P.E. (2016), "Earnings comparability as a signal of earnings quality and future stock returns: evidence from peer firms' earnings restatements", SSRN Working Paper No. 1966715, June.
	Chemmanur, T. and Yan, A. (2017), "Product market advertising, heterogeneous beliefs, and the long-run performance of initial public offerings", <i>Journal of Corporate Finance</i> , Vol. 46 No. 1, pp. 1-24.
	Chemmanur, T.J. and He, J. (2011), "IPO waves, product market competition, and the going public decision: theory and evidence", <i>Journal of Financial Economics</i> , Vol. 101 No. 2, pp. 382-412.
	Chen, C., Li, L. and Ma, M.L.Z. (2014), "Product market competition and the cost of equity capital: evidence from China", <i>Asia-Pacific Journal of Accounting & Economics</i> , Vol. 21 No. 3, pp. 37-41.
	Chen, C.X., Matsumura, E.M., Shin, J.Y. and Wu, S.Y. (2015), "The effect of competition intensity and competition type on the use of customer satisfaction measures in executive annual bonus contracts", <i>Accounting Review</i> , Vol. 90 No. 1, pp. 229-263.
	Chen, H., Lobo, G. and Wan, Y. (2010), "Association between borrower and lender state ownership and accounting conservatism", <i>Journal of Accounting Research</i> , Vol. 48 No. 5, pp. 973-1014.
	Chi, J.D. and Lee, D.S. (2010), "The conditional nature of the value of corporate governance", Journal of Banking & Finance, Vol. 34 No. 2, pp. 350-361.
	Chircop, J., Collins, D.W. and Hass, L.H. (2016), "Accounting comparability and corporate innovative efficiency", SSRN Working Paper No. 2810448, July.
	Choi, J. and Wong, T. (2007), "Auditors' governance functions and legal environments: an international investigation*", <i>Contemporary Accounting Research</i> , Vol. 24 No. 1, pp. 13-46.
	Chou, J., Ng, L., Sibilkov, V. and Wang, Q. (2011), "Product market competition and corporate governance", <i>Review of Development Finance</i> , Vol. 1 No. 2, pp. 114-130.
	Choi, JH., Choi, S., Myers, L.A. and Ziebart, D.A. (2017), "Financial statement comparability and the informativeness of stock prices about future earnings", SSRN Work Paper No. 2337571, February.
	Clemens, M.A., Redelet, S. and Bhavnani, R. (2012), "Counting chickens when they hatch: the short term effect of aid on growth", <i>Economic Journal</i> , Vol. 90 No. 561, pp. 1-33.
	Clinch, G. and Verrecchia, R.E. (1997), "Competitive disadvantage and discretionary disclosure in industries", <i>Australian Journal of Management</i> , Vol. 22 No. 2, pp. 125-137.
	Conway, P., Herdi, R., Chalaux, T., He, P. and Yu, J. (2010), "Product market regulation and competition in China", OECD Economics Department working papers, December.
	Cormier, D., Lapointe-Antunes, P. and Magnan, M. (2016), "CEO power and CEO hubris: a prelude to financial misreporting?", <i>Management Decision</i> , Vol. 54 No. 2, pp. 522-554.
	Cuadrado-Ballesteros, B., Garcia-Sanchez, IM. and Ferrero, J.M.F. (2016), "How are corporate disclosures related to the cost of capital? The fundamental role of information asymmetry",
	Management Decision, Vol. 54 No. 7, pp. 1669-1701.
للاستشارات	اطنارة

- Darrough, M.N. and Stoughton, N.M. (1990), "Financial disclosure policy in an entry game", Journal of Accounting and Economics, Vol. 12 Nos 1-3, pp. 219-243.
- Datta, S., Iskandar-Datta, M. and Sharma, V. (2011), "Product market pricing power, industry concentration and analysts' earnings forecasts", *Journal of Banking and Finance*, Vol. 35 No. 6, pp. 1352-1366.
- Datta, S., Iskandar-Datta, M. and Singh, V. (2013), "Product market power, industry structure, and corporate earnings management", *Journal of Banking & Finance*, Vol. 37 No. 8, pp. 3273-3285.
- De Franco, G., Kothari, S.P. and Verdi, R.S. (2011), "The benefits of financial statement comparability", Journal of Accounting Research, Vol. 49 No. 4, pp. 895-931.
- Dechow, P.M. and Dichev, I.D. (2002), "The quality of accruals and earnings: the role of accruals estimation errors", *The Accounting Review*, Vol. 77 No. 1, pp. 35-59.
- Dechow, P.M., Sloan, R.G. and Sweeny, A.P. (1995), "Detecting earnings management", *The Accounting Review*, Vol. 70 No. 2, pp. 193-225.
- Dechow, P.M., Sloan, R.G. and Sweeny, A.P. (1996), "Causes and consequences of earnings manipulation: an analysis of firms subject to enforcement actions by the SEC", *Contemporary Accounting Research*, Vol. 13 No. 1, pp. 1-36.
- Demsetz, H. and Lehn, K. (1985), "The structure of corporate ownership: causes and consequences", Journal of Political Economy, Vol. 93 No. 6, pp. 1155-1177.
- Dhaliwal, D., Huang, S., Khurana, I. and Pereira, R. (2014), "Product market competition and conditional conservatism", *Review of Accounting Studies*, Vol. 19 No. 4, pp. 1-37.
- Duchin, R., Matsusaka, J.G. and Ozbas, O. (2010), "When are outside directors effective?", *Journal of Financial Economics*, Vol. 96 No. 2, pp. 195-214.
- Evans, J.H. and Sridhar, S.S. (2002), "Disclosure-disciplining mechanisms capital markets, product markets, and shareholder litigation", *The Accounting Review*, Vol. 77 No. 3, pp. 595-626.
- Faccio, M. (2006), "Politically connected firms: can they squeeze the state?", American Economic Review, Vol. 96 No. 1, pp. 369-386.
- Fang, V.W., Iselin, M. and Zhang, G. (2017), "Financial statement comparability: benefits and costs", SSRN Working Paper No. 2858301, September.
- Financial Accounting Standards Board (FASB) (2010), "FASB (Financial Accounting Standards Board) statement of financial accounting concepts No. 8, Conceptual framework for financial reporting", Chapter 1, the Objective of General Purpose Financial Reporting, and Chapter 3, Qualitative Characteristics of Useful Information, Financial Accounting Standards Board (FASB).
- Francis, J.R. and Wang, D. (2008), "The joint effect of investor protection and big 4 audits on earnings quality around the world", *Contemporary Accounting Research*, Vol. 25 No. 1, pp. 157-191.
- Frésard, L. and Valta, P. (2011), "Competitive pressure and corporate policies", working paper, June, pp. 1-45.
- Gigler, F. (1994), "Self-enforcing voluntary disclosures", *Journal of Accounting Research*, Vol. 32 No. 2, p. 224.
- Grullon, G. and Michaely, R. (2007), "Corporate payout policy and product market competition", SSRN Working Paper No. 972221, March.
- Guadalupe, M. and Pérez-González, F. (2010), "Competition and private benefits of control", SSRN Working Paper No. 890814, October.
- Guo, Y., Jung, B. and Yang, S.Y. (2015), "Product market competition and earnings quality : a non linear relationship", SSRN Working Paper No. 2564777, February.
- Habib, A., Hassan, M.M. and Al-hadi, A. (2017), "Financial statement comparability and corporate cash holdings", *Journal of Contemporary Accounting & Economics*, Vol. 13 No. 3, pp. 304-321.
- Han, S.H., Kim, M., Lee, D.H. and Lee, S. (2014), "Information asymmetry, corporate governance, and shareholder wealth: evidence from unfaithful disclosures of Korean listed firms", *Asia-Pacific Journal of Financial Studies*, Vol. 43 No. 5, pp. 690-720.



2467

Managerial

decisions

MD 56.11	Harris, M.S. (1998), "The association between competition and managers' business segment reporting decisions", <i>Journal of Accounting Research</i> , Vol. 36 No. 1, pp. 111-128.
00,11	Hart, O.D. (1983), "The market mechanism as an incentives scheme", <i>The Bell Journal of Economics</i> , Vol. 14 No. 2, pp. 366-382.
	He, W. (2012), "Agency problems, product market competition and dividend policy in japan", <i>Accounting & Finance</i> , Vol. 52 No. 3, pp. 873-901.
2468	He, X., Wong, T.J. and Young, D. (2012), "Challenges for Implementation of fair value accounting in emerging markets: evidence from China", <i>Contemporary Accounting Research</i> , Vol. 29 No. 2, pp. 538-562.
	Healy, P.M. and Palepu, K.G. (2001), "Information asymmetry, corporate disclosure, and the capital markets: a review of the empirical disclosure literature", <i>Journal of Accounting and Economics</i> , Vol. 31 Nos 1-3, pp. 405-440.
	Hoberg, G. and Phillips, G. (2010), "Real and financial industry booms and busts", <i>Journal of Finance</i> , Vol. 65 No. 1, pp. 45-86.
	Holm, C. and Schøler, F. (2010), "Reduction of asymmetric information through corporate governance mechanisms – the importance of ownership dispersion and exposure toward the international capital market", <i>Corporate Governance</i> , Vol. 18 No. 1, pp. 32-47.
	Holmström, B. (1982), "Moral hazard in teams", Bell Journal of Economics, Vol. 11 No. 2, pp. 74-91.
	Holz, C.A. (2008), "China's economic growth 1978-2025: what we know today about china's economic growth tomorrow", World Development, Vol. 36 No. 10, pp. 1665-1691.
	Horn, H., Lang, H. and Lundgren, S. (1994), "Competition, long run contracts and internal inefficiencies in firms", <i>European Economic Review</i> , Vol. 38 No. 2, pp. 213-233.
	Jensen, M.C. and Meckling, W.H. (1976), "Theory of the firm: managerial behavior, agency costs and ownership structure", <i>Journal of Financial Economics</i> , Vol. 3 No. 4, pp. 305-360.
	Jiang, F. and Kim, K.A. (2015), "Corporate governance in China: a modern perspective", <i>Journal of Corporate Finance</i> , Vol. 32 No. 1, pp. 190-216.
	Jiang, F., Kim, K.A., Nofsinger, J.R. and Zhu, B. (2015), "Product market competition and corporate investment: evidence from China", <i>Journal of Corporate Finance</i> , Vol. 35 No. 1, pp. 196-210.
	Kanagaretnam, K., Lobo, G.J. and Whalen, D.J. (2007), "Does good corporate governance reduce information asymmetry around quarterly earnings announcements?", <i>Journal of Accounting and</i> <i>Public Policy</i> , Vol. 26 No. 4, pp. 497-522.
	Karuna, C. (2007), "Industry product market competition and managerial incentives", Journal of Accounting and Economics, Vol. 43 No. 2, pp. 275-297.
	Ke, B., Lennox, C.S. and Xin, Q. (2015), "The effect of China's weak institutional environment on the quality of big 4 audits", Accounting Review, Vol. 90 No. 4, pp. 1591-1619.
	Khanna, T. and Rivkin, J.W. (2001), "Estimating the performance effects of business groups in emerging markets", <i>Strategic Management Journal</i> , Vol. 22 No. 1, pp. 45-74.
	Kim, JB., Li, L., Lu, L.Y. and Yu, Y. (2016), "Financial statement comparability and expected crash risk", <i>Journal of Accounting and Economics</i> , Vol. 61 Nos 2-3, pp. 294-312.
	Kim, S., Kraft, P. and Ryan, S.G. (2013), "Financial statement comparability and credit risk", <i>Review of Accounting Studies</i> , Vol. 18 No. 3, pp. 783-823.
	Kim, S.M. and Kim, Y. (2017), "Product market competition on the effectiveness of internal control", <i>Asia-Pacific Journal of Accounting & Economics</i> , Vol. 24 Nos 1-2, pp. 163-182.
	Koopman, R.B., Wang, Z. and Wei, S.J. (2011), "How much of Chinese exports is really made in China? Assessing foreign and domestic value-added in gross exports", NBER Working Paper No. 14109, June.
	Kothari, S.P., Leone, A.J. and Wasley, C.E. (2005), "Performance matched discretionary accrual measures", <i>Journal of Accounting and Economics</i> , Vol. 39 No. 1, pp. 163-197.
للاستشارات	المنارة

www.

Laksmana, I. and Yang, Y. (2014), "Product market competition and earnings management: Evidence from discretionary accruals and real activity manipulation", *Advances in Accounting*, Vol. 30 No. 2, pp. 263-275.

- Lee, J. and Liu, X. (2016), "Competition, capital market feedback, and earnings management: evidence from economic deregulation", SSRN Working Paper No. 2348294, August.
- Leuz, C. (2003), "IAS versus U.S. GAAP: information asymmetry-based evidence from Germany's new market", *Journal of Accounting Research*, Vol. 41 No. 3, pp. 445-472.
- Leuz, C., Nanda, D. and Wysocki, P.D. (2003), "Earnings management and investor protection: an international comparison", *Journal of Financial Economics*, Vol. 69 No. 3, pp. 505-527.
- Li, X. (2010), "The impacts of product market competition on the quantity and quality of voluntary disclosures", *Review of Accounting Studies*, Vol. 15 No. 3, pp. 663-711.
- Lin, C., Officer, M.S. and Zhan, X. (2015), "Does competition affect earnings management? Evidence from a natural experiment", SSRN Working Paper No. 2357783, September.
- Majeed, M.A. and Zhang, X. (2016), "Competitive pressure and managerial decisions product market competition and earnings quality in China", *Chinese Management Studies*, Vol. 10 No. 3, pp. 559-592.
- Majeed, M.A., Zhang, X. and Umar, M. (2018), "Impact of investment efficiency on cost of equity: evidence from China", *Journal of Asia Business Studies*, Vol. 12 No. 1, pp. 44-59.
- Majeed, M.A., Zhang, X. and Wang, Z. (2017), "Product market competition, regulatory changes, ownership structure and accounting conservatism: evidence from China", *Chinese Management Studies*, Vol. 11 No. 4, pp. 658-688.
- Marciukaityte, D. and Park, J.C. (2009), "Market competition and earnings management", SSRN Working Paper No. 1361905, March.
- Martin, S. (1993), "Endogenous firm efficiency in a cournot principal-agent model", *Journal of Economic Theory*, Vol. 59 No. 2, pp. 445-450.
- Mnasri, K. and Ellouze, D. (2015), "Ownership structure, product market competition and productivity: evidence from Tunisia", *Management Decision*, Vol. 53 No. 8, pp. 1771-1805.
- Morrison, W.M. (2015), "China's economic rise: history, trends, challenges, and implications for the United States", Current Politics and Economics of Northern and Western Asia, Vol. 23 No. 4, pp. 493-559.
- Nalebuff, B.J. and Stiglitzs, J.E. (1983), "Prizes and incentives: towards a general theory of compensation and competition", *The Bell Journal of Economics*, Vol. 14 No. 1, pp. 21-43.
- Neel, M. (2017), "Accounting comparability and economic outcome of mandatory IFRS adoption", *Contemporary Accounting Research*, Vol. 34 No. 1, pp. 658-690.
- Newman, P. and Sansing, R. (1993), "Disclosure policies with multiple users", Journal of Accounting Research, Vol. 31 No. 1, pp. 92-112.
- Nickel, SJ. (1996), "Competition and corporate performance", Journal of Political Economy, Vol. 104 No. 4, pp. 724-746.
- Nickell, S., Wadhwani, S. and Wall, M. (1992), "Productivity growth in U.K. companies, 1975-1986", *European Economic Review*, Vol. 36 No. 5, pp. 1055-1085.
- Owen, B.M., Sun, S. and Zheng, W. (2008), "China's competition policy reforms : the anti-monopoly law and beyond", *Antitrust Law Journal*, Vol. 75 No. 1, pp. 231-265.
- Peng, M.W., Bruton, G.D., Stan, C.V. and Huang, Y. (2016), "Theories of the (state-owned) firm", Asia Pacific Journal of Management, Vol. 33, pp. 293-317.
- Peng, W.Q., Wei, K.C.J. and Yang, Z. (2011), "Tunneling or propping: evidence from connected transactions in China", *Journal of Corporate Finance*, Vol. 17 No. 2, pp. 306-325.



Managerial

decisions

Lang, M.H., Maffett, M.G. and Owens, E.L. (2010), "Earnings comovement and accounting comparability: the effects of mandatory IFRS adoption", SSRN Working Paper No. 1676937, September.

MD 56.11	Phillips, F., Libby, R. and Libby, P. (2013), <i>Fundamentals of Financial Accounting</i> , 4th ed., McGraw-Hill Irwin, New York, NY.					
	Piotroski, J.D. and Wong, T.J. (2012), "Institutions and information environment of Chinese listed firms", Fan, J. and Morck, R. (Eds), <i>Capitalizing China</i> , University of Chicago Press, Chicago, IL, pp. 201-24					
	Rotemberg, J.J. and Scharfstein, D.S. (1990), "Shareholder-value maximization and product-market competition", <i>The Review of Economic Studies</i> , Vol. 3 No. 3, pp. 367-391.					
2470	Scharfstein, D. (1988), "Product-market competition and managerial slack", <i>The RAND Journal of Economics</i> , Vol. 19 No. 1, pp. 147-155.					
	Schmalensee, R. (1978), "Entry deterrence in the ready-to-eat breakfast cereal industry.pdf", The Bell Journal of Economics, Vol. 9 No. 2, pp. 305-327.					
	Schmidt, K.M. (1997), "Managerial incentives and product market competition", The Review of Economic Studies, Vol. 64 No. 2, pp. 191-213.					
	Shailer, G. and Wang, K. (2015), "Government ownership and the cost of debt for Chinese listed corporations", <i>Emerging Markets Review</i> , Vol. 22 No. 1, pp. 1-17.					
	Shane, P., Smith, D. and Zhang, S. (2014), "Financial statement comparability and valuation of seasoned equity offerings", SSRN Working Paper No. 2372965, February.					
	Shao, Y., Hernández, R. and Liu, P. (2015), "Government intervention and corporate policies: Evidence from China", <i>Journal of Business Research</i> , Vol. 68 No. 6, pp. 1205-1215.					
	Shen, W. and Lin, C. (2009), "Firm profitability, state ownership, and top management turnover at the listed firms in China: a behavioral perspective", <i>Corporate Governance</i> , Vol. 17 No. 4, pp. 443-456.					
	Shleifer, A. (2004), "Does competition destroy ethical behavior?", <i>The American Economic Review</i> , Vol. 94 No. 2, pp. 414-418.					
	Stiebale, J. (2011), "Do financial constraints matter for foreign market entry? A firm-level examination", <i>World Economy</i> , Vol. 34 No. 1, pp. 123-153.					
	Stigler, G.J. (1958), "The economies of scale", Journal of Law and Economics, Vol. 1 No. 2, pp. 54-71.					
	Stigler, G.J. (1968), "Price and non-price competition", <i>Journal of Political Economy</i> , Vol. 76 No. 1, pp. 149-154.					
	Stivers, A.E. (2004), "Unraveling of information: competition and uncertainty", <i>The B.E. Journal of Theoretical Economics</i> , Vol. 4 No. 1, pp. 1-18.					
	Tybout, J.R. (2004), "Plant- and firm-level evidence on 'new' trade theories", in Choi, E.K. and Harrigan, J. (Eds), <i>Handbook of International Trade</i> , Blackwell Publishing, Oxford, pp. 388-415.					
	Valta, P. (2012), "Competition and the cost of debt", <i>Journal of Financial Economics</i> , Vol. 105 No. 3, pp. 661-682.					
	Verrecchia, R.E. and Weber, J. (2006), "Redacted disclosure", <i>Journal of Accounting Research</i> , Vol. 44 No. 4, pp. 791-814.					
	Verrecchia, R.E. (1983), "Discretionary disclosure", <i>Journal of Accounting & Economics</i> , Vol. 5 No. 1, pp. 179-194.					
	Verrecchia, R.E. (1990), "Endogenous proprietary costs through firm interdependence", Journal of Accounting & Economics, Vol. 12 No. 1, pp. 245-250.					
	Wang (2014), "Independent directors and corporate performance in China: a meta-empirical study", International Journal of Business Management, Vol. II No. 3, pp. 145-171.					
	Wang, B. and Xin, Q. (2011), "Auditor choice and accruals patterns of cross-listed firms", <i>China Journal of Accounting Research</i> , Vol. 4 No. 4, pp. 233-251.					
	Watts, R. and Zimmerman, J.L. (1986), Positive Accounting Theory, Prentice Hall, Englewood, NJ.					
	Wild, J.J. (1996), "The audit committee and earnings quality", <i>Journal of Accounting, Auditing and Finance</i> , Vol. 11 No. 2, pp. 247-276.					



Wu, P., Gao, L., Chen, Z. and Li, X. (2016), "Managing reputation loss in China: in-depth analyses of financial restatements", *Chinese Management Studies*, Vol. 10 No. 2, pp. 312-345.

Appendix

Descriptive Statist	ics							
-	Mean	Median	Max	Min	SD			
HHI	0.0177	0.0058	0.2171	0.0001	0.0344			
NF	81.1467	71	238	16	49			
C4	0.4437	0.4252	0.8541	0.1361	0.1566			
MKTS	26.175	26.2282	28.1077	23.2982	1.0069			
PPE	4.10E+07	5,755.246	1.04E+09	65,401.09	1.31E + 08			
CAPX	9356,262	1,522.270	2.26E + 08	19,390.51	2.88E+07			
ROA	-3.3245	-3.1722	-2.2265	-6.7945	0.7482			
PMC_EXST	0.2807	0.3183	3.0841	-4.3223	1.2319			
PMC_POTN	-0.0434	-0.3746	9.8347	-0.6708	1.2915			
Correlation matri	x							
	HHI	NF	C4	MKTS	PPE	CAPX	ROA	
HHI	1							
NF	-0.2777	1						
C4	0.5979	-0.5307	1					
MKTS	-0.1999	0.5007	-0.1152	1				
PPE	-0.2099	0.2499	-0.2027	0.2959	1			
CAPX	-0.2514	0.2475	-0.2255	0.2982	0.963	1		
ROA	-0.1168	0.1367	0.0624	0.0661	0.1754	0.1402	1	
PMC_EXST	-0.6052	0.8220	-0.3749	0.7528	0.2242	0.1318	0.0100	
PMC_POTN	0.1262	-0.1272	0.1167	-0.1384	-0.989	-0.9902	-0.1300	
Standardized scor	ing coefficient	S						
	HHI	NF	C4	MKTS	PPE	CAPX	ROA	
PMC_EXST	0.3968	-0.4722	0.4622	-0.2951				Table AI.
PMC_POTN				0.3325	0.6995	0.5984	-0.11	Summary of principle
Principal compon	ents are rotate	ed using orthog	gonal rotation t	technique				component analysis

Corresponding author

Chao Yan can be contacted at: chaoyan@zuel.edu.cn

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com

Managerial decisions

Zou, H.L., Zeng, S.X., Lin, H. and Xie, X.M. (2015), "Top executives' compensation, industrial competition, and corporate environmental performance", *Management Decision*, Vol. 53 No. 9, pp. 2036-2059.

Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

